# BIG IN REVERSE MATHEMATICS: THE UNCOUNTABILITY OF THE REALS

#### SAM SANDERS

**Abstract.** The uncountability of  $\mathbb{R}$  is one of its most basic properties, known far outside of mathematics. Cantor's 1874 proof of the uncountability of  $\mathbb{R}$  even appears in the very first paper on set theory, i.e., a historical milestone. In this paper, we study the uncountability of  $\mathbb{R}$  in Kohlenbach's *higher-order* Reverse Mathematics (RM for short), in the guise of the following principle:

for a countable set  $A \subset \mathbb{R}$ , there exists  $y \in \mathbb{R} \setminus A$ .

An important conceptual observation is that the usual definition of countable set—based on injections or bijections to  $\mathbb{N}$ —does not seem suitable for the RM-study of mainstream mathematics; we also propose a suitable (equivalent over strong systems) alternative definition of countable set, namely *union over*  $\mathbb{N}$ *of finite sets*; the latter is known from the literature and closer to how countable sets occur 'in the wild'. We identify a considerable number of theorems that are equivalent to the centred theorem based on our alternative definition. Perhaps surprisingly, our equivalent theorems involve most basic properties of the Riemann integral, regulated or bounded variation functions, Blumberg's theorem, and Volterra's early work circa 1881. Our equivalences are also *robust*, promoting the uncountability of  $\mathbb{R}$  to the status of 'big' system in RM.

## §1. Introduction.

**1.1. Summary.** Like Hilbert [34], we believe the infinite to be a central object of study in mathematics. That the infinite comes in 'different sizes' is a relatively new insight, due to Cantor around 1874 [19], in the guise of the *uncountability of*  $\mathbb{R}$ , also known simply as *Cantor's theorem*. We have previously studied the uncountability of  $\mathbb{R}$  in the guise of the following *third-order*<sup>1</sup> principles.

- NIN: there is no injection from [0, 1] to  $\mathbb{N}$ .
- NBI: there is no bijection from [0, 1] to  $\mathbb{N}$ .



Received August 1, 2022.

<sup>2020</sup> Mathematics Subject Classification. Primary 03B30, Secondary 03F35.

*Key words and phrases.* uncountability of  $\mathbb{R}$ , Reverse Mathematics, bounded variation, regulated, height function.

<sup>&</sup>lt;sup>1</sup>By definition, the uncountability of  $\mathbb{R}$  is formulated in terms of arbitrary (third-order) mappings from  $\mathbb{R}$  to  $\mathbb{N}$ . For this reason, we think it best to study this principle in a framework that directly includes such mappings, as opposed to representing them via second-order objects.

<sup>©</sup> The Author(s), 2023. Published by Cambridge University Press on behalf of The Association for Symbolic Logic. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution, and reproduction in any medium, provided the original work is properly cited.

In particular, as shown in [69–71], the principles NBI and NIN are *hard to prove* in terms of conventional<sup>2</sup> comprehension, while the objects claimed to exist are *hard to compute* in terms of the other data, in the sense of Kleene's computability theory based on S1–S9 [46, 58]. As shown in [79], this hardness remains if we restrict the mappings in NIN and NBI to well-known function classes, e.g., based on bounded variation, Borel, upper semi-continuity, and quasi-continuity. Moreover, *many* basic third-order theorems imply NIN or NBI, and the same at the computational level (see [68–71, 77–80]). Finally, NIN and NBI seem to be the weakest *natural* third-order principles that boast all the aforementioned properties.

For all these reasons, the study of the uncountability of  $\mathbb{R}$  in *Reverse Mathematics* (RM for short; see Section 1.3.1) seems like a natural enterprise. However, *try as we might* we have not managed to obtain elegant equivalences for NIN and NBI, working in Kohlenbach's higher-order *Reverse Mathematics* (see Section 1.3.1).

As argued in detail in Section 2, our main problem is that countable sets that occur 'in the wild' do not have injections (let alone bijections) to  $\mathbb{N}$  that can be defined in weak logical systems. By contrast, the (equivalent over ZF and weaker systems) definition of countable set as in Definitions 1.1 and 1.2 is much more suitable for the development of higher-order RM and is central to this paper.

DEFINITION 1.1. A set  $A \subset \mathbb{R}$  is *height countable* if there is a *height function*  $H : \mathbb{R} \to \mathbb{N}$  for A, i.e., for all  $n \in \mathbb{N}$ ,  $A_n := \{x \in A : H(x) < n\}$  is finite.

DEFINITION 1.2 (Finite set). Any  $X \subset \mathbb{R}$  is *finite* if there is  $N \in \mathbb{N}$  such that for any finite sequence  $(x_0, \ldots, x_N)$  of distinct reals, there is  $i \leq N$  such that  $x_i \notin X$ .

The notion of 'height function' can be found in the literature in connection to countability [40, 49, 59, 75, 97], while 'height countable' essentially amounts to *union* over  $\mathbb{N}$  of finite sets. By contrast, we believe Definition 1.2 has not been studied in the literature. Our move away from injections/bijections towards height functions and finite sets constitutes a 'shift of definition' which has ample historical precedent in RM and constructive mathematics, as discussed in Remark 2.1. We note that Kleene's quantifier ( $\exists^2$ ) from Section 1.3.2 is needed to make Definition 1.2 well-behaved, as discussed in more detail in Remark 3.5.

In more detail, we shall establish a large number of equivalences for the following principle, which is based on Definition 1.1 and expresses that [0, 1] is uncountable:

• NIN<sub>alt</sub>: the unit interval is not height countable.

In particular, we show in Section 3 that NIN<sub>alt</sub> is equivalent to the following natural principles, working in Kohlenbach's higher-order Reverse Mathematics, introduced in Section 1.3.1. Recall that a *regulated* function has left and right limits everywhere, as studied by Bourbaki for Riemann integration (see Section 1.3.3).

- (i) For regulated  $f : [0, 1] \to \mathbb{R}$ , there is a point  $x \in [0, 1]$  where f is continuous (or quasi-continuous, or lower semi-continuous, or Darboux).
- (ii) For regulated  $f : [0, 1] \to \mathbb{R}$ , the set of continuity points is dense in [0, 1].

<sup>&</sup>lt;sup>2</sup>We discuss the notion of 'conventional' comprehension in Section 1.3.2 where we introduce  $Z_2^{\omega}$ : a (conservative) higher-order extension of second-order arithmetic  $Z_2$  involving 'comprehension functionals'  $S_k^2$  that decide arbitrary  $\Pi_k^1$ -formulas. Since  $Z_2^{\omega}$  cannot prove NIN (see [69]) and both are essentially third-order in nature, our claim 'NIN is hard to prove' seems justified.

- (iii) For regulated  $f : [0, 1] \to [0, 1]$  with Riemann integral  $\int_0^1 f(x) dx = 0$ , there is  $x \in [0, 1]$  with f(x) = 0 (Bourbaki [10, p. 61, Corollary 1]).
- (iv) (Volterra [98]) For regulated  $f, g : [0, 1] \to \mathbb{R}$ , there is  $x \in [0, 1]$  such that f and g are both continuous or both discontinuous at x.
- (v) (Volterra [98]) For regulated  $f : [0, 1] \to \mathbb{R}$ , there is either  $q \in \mathbb{Q} \cap [0, 1]$ where f is discontinuous, or  $x \in [0, 1] \setminus \mathbb{Q}$  where f is continuous.
- (vi) For regulated  $f : [0, 1] \to \mathbb{R}$ , there is  $y \in (0, 1)$  where  $F(x) := \lambda x$ .  $\int_0^x f(t) dt$  is differentiable with derivative equal to f(y).
- (vii) For regulated  $f : [0,1] \to \mathbb{R}$ , there are  $a, b \in [0,1]$  such that  $\{x \in [0,1] : f(a) \le f(x) \le f(b)\}$  is infinite.
- (viii) Blumberg's theorem [8] restricted to regulated functions.

A full list of equivalences may be found in Section 3.3, while we obtain similar (but also very different) results for functions of *bounded variation* in Section 3.4. We introduce all required definitions in Section 1.3. Some of the above theorems, including items (iv) and (v) in the list, stem from Volterra's early work (1881) in the spirit of-but predating-the Baire category theorem, as discussed in Section 1.2.

Now, comparing items (i) and (ii) suggests that our results are *robust* as follows:

A system is *robust* if it is equivalent to small perturbations of itself. [60, p. 432; emphasis in original]

Most of our results shall be seen to exhibit a similar (or stronger) level of robustness. In this light, we feel that the uncountability of  $\mathbb{R}$  deserves the moniker 'big' system in the way this notion is used in second-order RM, namely as boasting many equivalences from various different fields of mathematics.

Next, items (i)–(viii) above imply NIN and are therefore *hard to prove* in the sense of Footnote 2. By contrast, we show in Section 3.5 that adding the extra condition 'Baire 1', makes these items provable from (essentially) arithmetical comprehension. While regulated functions are of course Baire 1, say over ZF, there is no contradiction here as the statement *a regulated function on the unit interval is Baire 1* already implies NIN (see [72, Section 2.8]). Other restrictions of items (i)–(viii), e.g., involving semicontinuity or Baire 2, are still equivalent to NIN<sub>alt</sub>, as shown in Section 3.3.3.

Finally, this paper deals with the RM of the uncountability of  $\mathbb{R}$ , while stronger 'completeness' properties of the reals, namely related to measure and category, are studied in [83]. In particular, the latter paper develops the higher-order RM of the *Baire category theorem* and Tao's *pigeon hole principle* for measure spaces [93]. We do not currently know of a principle weaker than the uncountability of  $\mathbb{R}$  that yields (interesting) RM-equivalences.

**1.2. Volterra's early work and related results.** We introduce Volterra's early work from [98] as it pertains to this paper, as well as related results.

First of all, the Riemann integral was groundbreaking for a number of reasons, including its ability to integrate functions with infinitely many points of discontinuity, as shown by Riemann himself [74]. A natural question is then 'how discontinuous' a Riemann integrable function can be. In this context, Thomae introduced the function  $T : \mathbb{R} \to \mathbb{R}$  around 1875 in [95, p. 14, Section 20]:

$$T(x) := \begin{cases} 0, & \text{if } x \in \mathbb{R} \setminus \mathbb{Q}, \\ \frac{1}{q}, & \text{if } x = \frac{p}{q} \text{ and } p, q \text{ are co-prime.} \end{cases}$$
(1.1)

Thomae's function *T* is integrable on any interval, but has a dense set of points of discontinuity, namely  $\mathbb{Q}$ , and a dense set of points of continuity, namely  $\mathbb{R} \setminus \mathbb{Q}$ .

The perceptive student, upon seeing Thomae's function as in (1.1), will ask for a function continuous at each rational point and discontinuous at each irrational one. Such a function cannot exist, as is generally proved using the Baire category theorem. However, Volterra in [98] already established this negative result about 20 years before the publication of the Baire category theorem.

Secondly, as to the content of Volterra's paper [98], we find the following theorem on the first page, where a function is *pointwise discontinuous* if it has a dense set of continuity points.

THEOREM 1.3 (Volterra, 1881). There do not exist pointwise discontinuous functions defined on an interval for which the continuity points of one are the discontinuity points of the other, and vice versa.

Volterra then states two corollaries, of which the following is perhaps well-known in 'popular mathematics' and constitutes the aforementioned negative result.

COROLLARY 1.4 (Volterra, 1881). There is no  $\mathbb{R} \to \mathbb{R}$ -function that is continuous on  $\mathbb{Q}$  and discontinuous on  $\mathbb{R} \setminus \mathbb{Q}$ .

Thirdly, we shall study Volterra's theorem and corollary restricted to regulated functions (see Section 1.3.3). The latter kinds of functions are automatically 'pointwise discontinuous' in the sense of Volterra.

Fourth, Volterra's results from [98] are generalised in [30, 86]. The following theorem is immediate from these generalisations.

**THEOREM 1.5.** For any countable dense set  $D \subset [0, 1]$  and  $f : [0, 1] \rightarrow \mathbb{R}$ , *f* is either *discontinuous at some point in D or continuous at some point in*  $[0, 1] \setminus D$ .

Perhaps surprisingly, this generalisation (restricted to bounded variation or regulated functions) is still equivalent to the uncountability of  $\mathbb{R}$ . The same holds for the related *Blumberg's theorem* with the same restrictions.

THEOREM 1.6 (Blumberg's theorem [8]). For any  $f : \mathbb{R} \to \mathbb{R}$ , there is a dense subset  $D \subset \mathbb{R}$  such that the restriction of f to D, usually denoted  $f_{\uparrow D}$ , is continuous.

To be absolutely clear, the conclusion of Blumberg's theorem means that

$$(\forall x \in D, \varepsilon > 0)(\exists \delta > 0)(\forall y \in D)(|x - y| < \delta \rightarrow |f(x) - f(y)| < \varepsilon)),$$

where the underlined quantifier marks the difference with 'usual' continuity.

**1.3. Preliminaries and definitions.** We briefly introduce *Reverse Mathematics* in Section 1.3.1. We introduce some essential axioms (Section 1.3.2) and definitions (Section 1.3.3). A full introduction may be found in, e.g., [69, Section 2].

*1.3.1. Reverse Mathematics.* Reverse Mathematics (RM hereafter) is a program in the foundations of mathematics initiated around 1975 by Friedman [28, 29] and developed extensively by Simpson [89]. The aim of RM is to identify the minimal axioms needed to prove theorems of ordinary, i.e., non-set theoretical, mathematics.

1611

We refer to [91] for a basic introduction to RM and to [26, 88, 89] for an overview of RM. We expect basic familiarity with RM, in particular Kohlenbach's *higherorder* RM [48] essential to this paper, including the base theory  $\text{RCA}_0^{\omega}$ . An extensive introduction can be found in, e.g., [65, 67–69]. All undefined notions may be found in [68, 69], while we do point out here that we shall sometimes use common notations from the type theory. For instance, the natural numbers are type 0 objects, denoted  $n^0$  or  $n \in \mathbb{N}$ . Similarly, elements of Baire space are type 1 objects, denoted  $f \in \mathbb{N}^{\mathbb{N}}$ or  $f^1$ . Mappings from Baire space  $\mathbb{N}^{\mathbb{N}}$  to  $\mathbb{N}$  are denoted  $Y : \mathbb{N}^{\mathbb{N}} \to \mathbb{N}$  or  $Y^2$ .

1.3.2. Some comprehension functionals. In second-order RM, the logical hardness of a theorem is measured via what fragment of the comprehension axiom is needed for a proof. For this reason, we introduce some axioms and functionals related to higher-order comprehension in this section. We are mostly dealing with conventional comprehension here, i.e., only parameters over  $\mathbb{N}$  and  $\mathbb{N}^{\mathbb{N}}$  are allowed in formula classes like  $\Pi_k^1$  and  $\Sigma_k^1$ .

First of all, the functional  $\varphi$  in  $(\exists^2)$  is also *Kleene's quantifier*  $\exists^2$  and is clearly discontinuous at  $f = 11 \dots$  in Cantor space:

$$(\exists \varphi^2 \leq_2 1) (\forall f^1) [(\exists n^0) (f(n) = 0) \leftrightarrow \varphi(f) = 0]. \tag{3}$$

In fact,  $(\exists^2)$  is equivalent to the existence of  $F : \mathbb{R} \to \mathbb{R}$  such that F(x) = 1 if  $x >_{\mathbb{R}} 0$ , and 0 otherwise (see [48, Proposition 3.12]). Related to  $(\exists^2)$ , the functional  $\mu^2$  in  $(\mu^2)$  is called *Feferman's*  $\mu$  (see [2]) and may be found—with the same symbol—in Hilbert-Bernays' Grundlagen [35, Supplement IV]:

$$\begin{aligned} (\exists \mu^2)(\forall f^1) \big[ (\exists n)(f(n) = 0) \to [f(\mu(f)) = 0 \land (\forall i < \mu(f))(f(i) \neq 0)] & (\mu^2) \\ \land [(\forall n)(f(n) \neq 0) \to \mu(f) = 0] \big]. \end{aligned}$$

We have  $(\exists^2) \leftrightarrow (\mu^2)$  over  $\mathsf{RCA}_0^{\omega}$  (see [48, Section 3]) and  $\mathsf{ACA}_0^{\omega} \equiv \mathsf{RCA}_0^{\omega} + (\exists^2)$  proves the same sentences as  $\mathsf{ACA}_0$  by [38, Theorem 2.5].

Secondly, the functional  $S^2$  in  $(S^2)$  is called *the Suslin functional* [48]:

$$(\exists \mathsf{S}^2 \leq_2 1)(\forall f^1) [(\exists g^1)(\forall n^0)(f(\overline{g}n) = 0) \leftrightarrow \mathsf{S}(f) = 0]. \tag{S}^2$$

The system  $\Pi_1^1$ -CA<sub>0</sub><sup> $\omega$ </sup> = RCA<sub>0</sub><sup> $\omega$ </sup> + (S<sup>2</sup>) proves the same  $\Pi_3^1$ -sentences as  $\Pi_1^1$ -CA<sub>0</sub> by [76, Theorem 2.2]. By definition, the Suslin functional S<sup>2</sup> can decide whether a  $\Sigma_1^1$ -formula as in the left-hand side of (S<sup>2</sup>) is true or false. We similarly define the functional S<sup>2</sup><sub>k</sub> which decides the truth or falsity of  $\Sigma_k^1$ -formulas from L<sub>2</sub>; we also define the system  $\Pi_k^1$ -CA<sub>0</sub><sup> $\omega$ </sup> as RCA<sub>0</sub><sup> $\omega$ </sup> + (S<sup>2</sup><sub>k</sub>), where (S<sup>2</sup><sub>k</sub>) expresses that S<sup>2</sup><sub>k</sub> exists. We note that the operators  $v_n$  from [18, p. 129] are essentially S<sup>2</sup><sub>n</sub> strengthened to return a witness (if existent) to the  $\Sigma_n^1$ -formula at hand.

Thirdly, full second-order arithmetic  $Z_2$  is readily derived from  $\cup_k \Pi_k^1$ -CA<sub>0</sub><sup> $\omega$ </sup>, or from:

$$(\exists E^3 \leq_3 1)(\forall Y^2) [(\exists f^1)(Y(f) = 0) \leftrightarrow E(Y) = 0], \qquad (\exists^3)$$

and we therefore define  $Z_2^{\Omega} \equiv \mathsf{RCA}_0^{\omega} + (\exists^3)$  and  $Z_2^{\omega} \equiv \bigcup_k \Pi_k^1 - \mathsf{CA}_0^{\omega}$ , which are conservative over  $Z_2$  by [38, Corollary 2.6]. Despite this close connection,  $Z_2^{\omega}$  and  $Z_2^{\Omega}$  can behave quite differently, as discussed in, e.g., [65, Section 2.2]. The functional from  $(\exists^3)$  is also called ' $\exists^3$ ', and we use the same convention for other functionals.

1.3.3. Some basic definitions. We introduce some definitions needed in the below, mostly stemming from mainstream mathematics. We note that subsets of  $\mathbb{R}$  are given by their characteristic functions as in Definition 1.7, well-known from measure and probability theory.

Zeroth of all, we make use the usual definition of (open) set, where B(x, r) is the open ball with radius r > 0 centred at  $x \in \mathbb{R}$ .

DEFINITION 1.7 (Sets).

- A subset A ⊂ ℝ is given by its characteristic function F<sub>A</sub> : ℝ → {0,1}, i.e., we write x ∈ A for F<sub>A</sub>(x) = 1, for any x ∈ ℝ.
- A subset  $O \subset \mathbb{R}$  is *open* in case  $x \in O$  implies that there is  $k \in \mathbb{N}$  such that  $B(x, \frac{1}{2^k}) \subset O$ .
- A subset  $C \subset \mathbb{R}$  is *closed* if the complement  $\mathbb{R} \setminus C$  is open.

As discussed in Remark 2.7, the study of functions of bounded variation already gives rise to open sets that do not come with additional representation beyond Definition 1.7.

First of all, we shall study the following notions of weak continuity, all of which hark back to the days of Baire, Darboux, and Volterra [3, 4, 22, 98].

DEFINITION 1.8. For  $f : [0, 1] \to \mathbb{R}$ , we have the following definitions:

- f is upper semi-continuous at  $x_0 \in [0, 1]$  if  $f(x_0) \ge_{\mathbb{R}} \limsup_{x \to x_0} f(x)$ .
- *f* is lower semi-continuous at  $x_0 \in [0, 1]$  if  $f(x_0) \leq_{\mathbb{R}} \liminf_{x \to x_0} f(x)$ .
- *f* is quasi-continuous at  $x_0 \in [0, 1]$  if for  $\varepsilon > 0$  and an open neighbourhood *U* of  $x_0$ , there is a non-empty open  $G \subset U$  with  $(\forall x \in G)(|f(x_0) f(x)| < \varepsilon)$ .
- $f : \mathbb{R} \to \mathbb{R}$  symmetrically continuous at  $x \in \mathbb{R}$  if

$$(\forall \varepsilon > 0)(\exists \delta > 0)(\forall z \in \mathbb{R})(|z| < \delta \rightarrow |f(x+z) - f(x-z)| < \varepsilon).$$

- f is Baire 1 if it is the pointwise limit of a sequence of continuous functions.
- f is Baire 2 if it is the pointwise limit of a sequence of Baire 1 functions.
- f is *Baire*  $I^*$  if<sup>3</sup> there is a sequence of closed sets  $(C_n)_{n \in \mathbb{N}}$  such  $[0, 1] = \bigcup_{n \in \mathbb{N}} C_n$ and  $f_{\uparrow C_m}$  is continuous for all  $m \in \mathbb{N}$ .

The first two items are often abbreviated as 'usco' and 'lsco'.

Secondly, we also need the notion of 'intermediate value property', also called the 'Darboux property' in light of Darboux's work in [22].

DEFINITION 1.9 (Darboux property). Let  $f : [0, 1] \to \mathbb{R}$  be given.

- A real  $y \in \mathbb{R}$  is a left (resp. right) *cluster value* of f at  $x \in [0, 1]$  if there is  $(x_n)_{n \in \mathbb{N}}$  such that  $y = \lim_{n \to \infty} f(x_n)$  and  $x = \lim_{n \to \infty} x_n$  and  $(\forall n \in \mathbb{N})(x_n \le x)$  (resp.  $(\forall n \in \mathbb{N})(x_n \ge x)$ ).
- A point  $x \in [0, 1]$  is a *Darboux point* of  $f : [0, 1] \to \mathbb{R}$  if for any  $\delta > 0$  and any left (resp. right) cluster value y of f at x and  $z \in \mathbb{R}$  strictly between y and f(x), there is  $w \in (x \delta, x)$  (resp.  $w \in (x, x + \delta)$ ) such that f(w) = y.

<sup>&</sup>lt;sup>3</sup>The notion of Baire 1\* goes back to [27] and equivalent definitions may be found in [45]. In particular, Baire 1\* is equivalent to the Jayne–Rogers notion of *piecewise continuity* from [41].

By definition, a point of continuity is also a Darboux point, but not vice versa. Thirdly, we introduce the 'usual' definitions of countable set (Definitions 1.10 and 1.11).

DEFINITION 1.10 (Enumerable sets of reals). A set  $A \subset \mathbb{R}$  is *enumerable* if there exists a sequence  $(x_n)_{n \in \mathbb{N}}$  such that  $(\forall x \in \mathbb{R})(x \in A \to (\exists n \in \mathbb{N})(x =_{\mathbb{R}} x_n))$ .

This definition reflects the RM-notion of 'countable set' from [89, Theorem V.4.2]. We note that given  $\mu^2$  from Section 1.3.2, we may replace the final implication in Definition 1.10 by an equivalence.

DEFINITION 1.11 (Countable subset of  $\mathbb{R}$ ). A set  $A \subset \mathbb{R}$  is *countable* if there exists  $Y : \mathbb{R} \to \mathbb{N}$  such that  $(\forall x, y \in A)(Y(x) =_0 Y(y) \to x =_{\mathbb{R}} y)$ . The mapping  $Y : \mathbb{R} \to \mathbb{N}$  is called an *injection* from A to  $\mathbb{N}$  or *injective on* A. If  $Y : \mathbb{R} \to \mathbb{N}$  is also *surjective*, i.e.,  $(\forall n \in \mathbb{N})(\exists x \in A)(Y(x) = n)$ , we call A strongly countable.

The first part of Definition 1.11 is from Kunen's set theory textbook [54, p. 63] and the second part is taken from Hrbacek–Jech's set theory textbook [36] (where the term 'countable' is used instead of 'strongly countable'). For the rest of this paper, 'strongly countable' and 'countable' shall exclusively refer to Definition 1.11, *except when explicitly stated otherwise*.

Finally, the uncountability of  $\mathbb{R}$  can be studied in numerous guises in higher-order RM. For instance, the following are from [68, 69], where it is also shown that many extremely basic theorems imply these principles, while  $Z_2^{\omega}$  cannot prove them.

- For a countable set  $A \subset [0, 1]$ , there is  $y \in [0, 1] \setminus A$ .
- NIN: there is no injection from [0, 1] to  $\mathbb{N}$ .
- For a *strongly* countable set  $A \subset [0, 1]$ , there is  $y \in [0, 1] \setminus A$ .
- NBI: there is no *bijection* from [0, 1] to  $\mathbb{N}$ .

The reader will verify that the first two and last two items are (trivially) equivalent. Besides these and similar variations in [79], we have not been able to obtain elegant or natural equivalences involving the uncountability of  $\mathbb{R}$  try as we might. As discussed in Section 2, this is because the above items are formulated using the 'set theoretic' definition of countability as in Definition 1.11. In Section 3, we obtain many equivalences involving the uncountability of  $\mathbb{R}$ , based on the alternative (but equivalent over ZF) notion of 'height countable' introduced in Section 1.1.

*1.3.4. Some advanced definitions: bounded variation and around.* We formulate the definitions of bounded variation and regulated functions, and some background.

Firstly, the notion of *bounded variation* (often abbreviated *BV* below) was first explicitly<sup>4</sup> introduced by Jordan around 1881 [42] yielding a generalisation of Dirichlet's convergence theorems for Fourier series. Indeed, Dirichlet's convergence results are restricted to functions that are continuous except at a finite number of points, while *BV*-functions can have infinitely many points of discontinuity, as already studied by Jordan, namely in [42, p. 230]. Nowadays, the *total variation* of a function  $f : [a, b] \rightarrow \mathbb{R}$  is defined as follows:

$$V_a^b(f) := \sup_{a \le x_0 < \dots < x_n \le b} \sum_{i=0}^n |f(x_i) - f(x_{i+1})|.$$
(1.2)

<sup>&</sup>lt;sup>4</sup>Lakatos in [56, p. 148] claims that Jordan did not invent or introduce the notion of bounded variation in [42], but rather discovered it in Dirichlet's 1829 paper [23].

If this quantity exists and is finite, one says that f has bounded variation on [a, b]. Now, the notion of bounded variation is defined in [64] *without* mentioning the supremum in (1.2); this approach can also be found in [11, 12, 53]. Hence, we shall distinguish between the two notions in Definition 1.12. As it happens, Jordan seems to use item (a) of Definition 1.12 in [42, pp. 228–229]. This definition suggests a twofold variation for any result on functions of bounded variation, namely depending on whether the supremum (1.2) is given, or only an upper bound on the latter.

DEFINITION 1.12 (Variations on variation).

- (a) The function  $f : [a, b] \to \mathbb{R}$  has bounded variation on [a, b] if there is  $k_0 \in \mathbb{N}$  such that  $k_0 \ge \sum_{i=0}^{n} |f(x_i) f(x_{i+1})|$  for any partition  $x_0 = a < x_1 < \cdots < x_{n-1} < x_n = b$ .
- (b) The function  $f : [a, b] \to \mathbb{R}$  has a variation on [a, b] if the supremum in (1.2) exists and is finite.

Secondly, the fundamental theorem about BV-functions is as follows; this theorem is proved by Jordan in [42].

THEOREM 1.13 (Jordan decomposition theorem [42, p. 229]). A BV-function f: [0,1]  $\rightarrow \mathbb{R}$  is the difference of two non-decreasing functions g, h: [0,1]  $\rightarrow \mathbb{R}$ .

Theorem 1.13 has been studied via second-order representations in [31, 53, 64, 105]. The same holds for constructive analysis by [11, 12, 33, 73], involving different (but related) constructive enrichments. Now, ACA<sub>0</sub> suffices to derive Theorem 1.13 for various kinds of second-order *representations* of *BV*-functions in [53, 64]. By contrast, our results in [68] imply that the third-order version of Theorem 1.13 is hard to prove in terms of conventional comprehension.

Thirdly, Jordan proves in [43, Section 105] that BV-functions are exactly those for which the notion of 'length of the graph of the function' makes sense. In particular,  $f \in BV$  if and only if the 'length of the graph of f', defined as follows:

$$L(f, [0, 1]) := \sup_{0 \le t_0 \le t_1 \le \dots \le t_m = 1} \sum_{i=0}^{m-1} \sqrt{(t_i - t_{i+1})^2 + (f(t_i) - f(t_{i+1}))^2}$$
(1.3)

exists and is finite by [1, Theorem 3.28(c)]. In case the supremum in (1.3) exists (and is finite), *f* is also called *rectifiable*. Rectifiable curves predate *BV*-functions: in [84, Section 1 and 2], it is claimed that (1.3) is essentially equivalent to Duhamel's 1866 approach from [25, Chapter VI]. Around 1833, Dirksen, the PhD supervisor of Jacobi and Heine, already provides a definition of arc length that is (very) similar to (1.3) (see [24, Section 2, p. 128], but with some conceptual problems as discussed in [20, Section 3].

Fourth, a function is *regulated* (called 'regular' in [1]) if for every  $x_0$  in the domain, the 'left' and 'right' limits  $f(x_0 -) = \lim_{x \to x_0-} f(x)$  and  $f(x_0+) = \lim_{x \to x_0+} f(x)$ exist. Scheeffer studies discontinuous regulated functions in [84] (without using the term 'regulated'), while Bourbaki develops Riemann integration based on regulated functions in [9]. We note that *BV*-functions are regulated, while Weierstrass' 'monster' function is a natural example of a regulated function not in *BV*.

Finally, an interesting observation about regulated functions is as follows.

REMARK 1.14 (Continuity and regulatedness). First of all, as discussed in [48, Section 3], the *local* equivalence for functions on Baire space between sequential

and 'epsilon-delta' continuity cannot be proved in ZF. By [68, Theorem 3.32], this equivalence for *regulated* functions is provable in ZF (and actually just  $ACA_0^{\omega}$ ).

Secondly,  $\mu^2$  readily computes the left and right limits of regulated  $f : [0, 1] \rightarrow \mathbb{R}$ . In this way, the formula 'f is continuous at  $x \in [0, 1]$ ' is decidable using  $\mu^2$ , namely equivalent to the formula 'f(x+) = f(x) = f(x-)'. The usual 'epsilon-delta' definition of continuity involves quantifiers over  $\mathbb{R}$ , i.e., the previous equality is much simpler and more elementary.

By the previous remark, the basic notions needed for the study of regulated and BV-functions make sense in ACA<sub>0</sub><sup> $\omega$ </sup>.

§2. Countability by any other name. We show that the 'standard' set-theoretic definitions of countability-from Section 1.3.3 and based on injections and bijections to  $\mathbb{N}$ -are not suitable for the RM-study of regulated functions (see Section 2.1) and *BV*-functions (Section 2.2). We also formulate an alternative-more suitable for RM-notion of countability (Definitions 2.3 and 2.6), which amounts to 'unions over  $\mathbb{N}$  of finite sets' and which can also be found in the mathematical literature. This kind of 'shift of definition' has historical precedent as follows.

REMARK 2.1. First of all, the correct choice of definition for a given mathematical notion is crucial to the development of RM, as can be gleaned from the following quote from [15, p. 129].

Under the old definition [of real number from [87]], it would be consistent with RCA<sub>0</sub> that there exists a sequence of real numbers  $(x_n)_{n \in \mathbb{N}}$  such that  $(x_n + \pi)_{n \in \mathbb{N}}$  is not a sequence of real numbers. We thank Ian Richards for pointing out this defect of the old definition. Our new definition [of real number from [15]], given above, is adopted in order to remove this defect. All of the arguments and results of [87] remain correct under the new definition.

In short, the early definition of 'real number' from [87] was not suitable for the development of RM, highlighting the importance of the 'right' choice of definition.

Secondly, we stress that RM is not unique in this regard: the early definition of 'continuous function' in Bishop's constructive analysis [7] was also deemed problematic and changed to a new definition to be found in [13]; the (substantial) problems with both definitions are discussed in some detail in [14, 99], including elementary properties such as the concatenation of continuous functions and the continuity of  $\frac{1}{x}$  for x > 0.

In short, the development of mathematics in logical systems with 'restricted' resources, like RM or constructive mathematics, seems to hinge on the 'right' choice of definition. In this section, we argue that the 'right' definition of countability for higher-order RM is given by height functions as in Section 1.1. To be absolutely clear, the background theory for this section is ZFC, i.e., a statement like ' $A \subset \mathbb{R}$  is countable' means that the latter is provable in the former; most arguments (should) go through in  $\mathbb{Z}_2^{\Omega}$ .

**2.1. Regulated functions and countability.** As suggested in Section 1.1, the settheoretic definition of countable set is not suitable for the RM-study of regulated functions. We first provide some motivation for this claim in Remark 2.2. Inspired by the latter, we can then present our alternative notion in Definition 2.3, which amounts to 'unions over  $\mathbb{N}$  of finite sets'.

REMARK 2.2 (Countable sets by any other name). First of all, we have previously investigated the RM of regulated functions in [68]. As part of this study, the following sets—definable via  $\exists^2$ —present themselves, where  $f : [0, 1] \rightarrow \mathbb{R}$  is regulated:

$$A := \left\{ x \in (0,1) : f(x+) \neq f(x) \lor f(x-) \neq f(x) \right\},\$$
  
$$A_n := \left\{ x \in (0,1) : |f(x+) - f(x)| > \frac{1}{2^n} \lor |f(x-) - f(x)| > \frac{1}{2^n} \right\}.$$
 (2.1)

Clearly,  $A = \bigcup_{n \in \mathbb{N}} A_n$  collects all points in (0, 1) where f is discontinuous; this set is central to many proofs involving regulated functions (see, e.g., [1, Theorem 0.36]). Now, that  $A_n$  is finite follows by a standard<sup>5</sup> compactness argument. However, while A is then countable, we are unable to construct an injection from A to  $\mathbb{N}$  (let alone a bijection), even working in  $\mathbb{Z}_2^{\omega}$  (see Remark 2.7 for details).

In short, one readily finds countable sets 'in the wild', namely pertaining to regulated functions, for which the associated injections to  $\mathbb{N}$  cannot be constructed in reasonably weak logical systems.

Secondly, in light of (2.1), regulated functions give rise to countable sets given *only* as the union over  $\mathbb{N}$  of finite sets (i.e., without information about an injection to  $\mathbb{N}$ ). To see that the 'reverse' is also true, consider the following function:

$$h(x) := \begin{cases} 0, & x \notin \bigcup_{m \in \mathbb{N}} X_m, \\ \frac{1}{2^{n+1}}, & x \in X_n \text{ and } n \text{ is the least such number,} \end{cases}$$
(2.2)

where  $(X_n)_{n \in \mathbb{N}}$  is a sequence of finite sets in [0, 1]. One readily shows that *h* is regulated using  $\exists^2$ . For general closed sets, (2.2) is crucial to the study of Baire 1 functions (see [63, p. 238]). Hence, regulated functions yield countable sets given (only) as unions over  $\mathbb{N}$  of finite sets, namely via  $A = \bigcup_{n \in \mathbb{N}} A_n$  from (2.1), and vice versa, namely via  $h : [0, 1] \to \mathbb{R}$  as in (2.2).

In summary, we observe that the usual definition of countable set (involving injections/bijections to  $\mathbb{N}$ ) is not suitable for the RM-study of regulated functions. Luckily, (2.1) and (2.2) suggest an alternative approach via the fundamental connection between regulated functions on one hand, and countable sets given as

## the union over $\mathbb{N}$ of finite sets

on the other hand. In conclusion, the RM-study of regulated functions should be based on the centred notion of countability and *not* injections/bijections to  $\mathbb{N}$ .

Motivated by Remark 2.2, we introduce our alternative definition of countability, which is exactly the same as Definition 1.1 in Section 1.1.

DEFINITION 2.3. A set  $A \subset \mathbb{R}$  is *height countable* if there is a *height* function  $H : \mathbb{R} \to \mathbb{N}$  for A, i.e., for all  $n \in \mathbb{N}$ ,  $A_n := \{x \in A : H(x) < n\}$  is finite.

<sup>&</sup>lt;sup>5</sup>If  $A_n$  were infinite, the Bolzano–Weierstrass theorem implies the existence of a limit point  $y \in [0, 1]$  for  $A_n$ . One readily shows that f(y+) or f(y-) does not exist, a contradiction as f is assumed to be regulated.

The previous notion of 'height' is mentioned in the context of countability in, e.g., [40, 49, 59, 75, 97]. Definition 2.3 amounts to 'union over  $\mathbb{N}$  of finite sets', as is readily shown in ACA<sub>0</sub><sup>0</sup>.

Finally, the observations from Remark 2.2 regarding countable sets also apply *mutatis mutandis* to finite sets. Indeed, finite as each  $A_n$  from (2.1) may be, we are unable to construct an injection to a finite subset of  $\mathbb{N}$ , even assuming  $\mathbb{Z}_2^{\omega}$  (see Remark 2.7 for details). By contrast, the definition of finite set from Section 1.1 is more suitable: one readily<sup>6</sup> shows that  $A_n$  from (2.1) is finite as in Definition 2.4, which is exactly the same as Definition 1.2 in Section 1.1.

DEFINITION 2.4 (Finite set). Any  $X \subset \mathbb{R}$  is *finite* if there is  $N \in \mathbb{N}$  such that for any finite sequence  $(x_0, ..., x_N)$  of distinct reals, there is  $i \leq N$  such that  $x_i \notin X$ .

The number N from Definition 2.4 is call a *size bound* for the finite set  $X \subset \mathbb{R}$ . Analogous to countable sets, the RM-study of regulated functions should be based on Definition 2.4 and *not* on the set-theoretic definition based on injections/bijections to finite subsets of  $\mathbb{N}$  or similar constructs.

**2.2. Bounded variation functions and countability.** We discuss the observations from Section 2.1 for the particular case of functions of bounded variation (which are regulated by Theorem 3.3). In particular, while the same observations apply, they have to be refined to yield elegant equivalences.

**REMARK** 2.5 (Countable by another name). First of all, we consider (2.1), but formulated for a *BV*-function  $g : [0, 1] \to \mathbb{R}$ , as follows:

$$B := \left\{ x \in (0,1) : g(x+) \neq g(x) \lor g(x-) \neq g(x) \right\},\$$
  

$$B_n := \left\{ x \in (0,1) : |g(x+) - g(x)| > \frac{1}{2^n} \lor |g(x-) - g(x)| > \frac{1}{2^n} \right\}.$$
(2.3)

Similar to  $A = \bigcup_{n \in A_n}$  as in (2.3),  $B = \bigcup_{n \in \mathbb{N}} B_n$  collects all points in (0, 1) where g is discontinuous and this set is central to many proofs involving *BV*-functions (see [1]). Similar to A from (2.3), B is countable but we are unable to construct an injection from B to  $\mathbb{N}$  (let alone a bijection), even assuming  $\mathbb{Z}_2^{\infty}$  (see Remark 2.7).

Secondly, there is a crucial difference between (2.1) and (2.3): we know that the set  $B_n$  is finite and has at most  $2^n V_0^1(f)$  elements; indeed, each element of  $B_n$  contributes at least  $1/2^n$  to the total variation  $V_0^1(f)$  as in (1.2). By contrast, we have no extra information about the size of  $A_n$  from (2.1). However, this extra information is crucial if we wish to deal with BV-functions (only). Indeed, the function h from (2.2) is not in BV, e.g., in the trivial case where each  $X_n$  has at least  $2^{n+1}$  elements. By contrast, consider the following nicer function:

$$k(x) := \begin{cases} 0, & x \notin \bigcup_{m \in \mathbb{N}} Y_m, \\ \frac{1}{2^{n+1}} \frac{1}{g(n)+1}, & x \in Y_n \text{ and } n \text{ is the least such number,} \end{cases}$$
(2.4)

where  $g \in \mathbb{N}^{\mathbb{N}}$  is a *width function*<sup>7</sup> for  $(Y_n)_{n \in \mathbb{N}}$ . One readily verifies that  $k : [0, 1] \to \mathbb{R}$  is in *BV* with total variation bounded by 1. Hence, *BV*-functions yield countable

<sup>&</sup>lt;sup>6</sup>The proof of Theorem 3.6 shows that  $A_n$  is finite, working in ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup>.

<sup>&</sup>lt;sup>7</sup>The function  $g \in \mathbb{N}^{\mathbb{N}}$  is a *width function* for the sequence of sets  $(Y_n)_{n \in \mathbb{N}}$  in  $\mathbb{R}$  in case  $Y_n$  has at most g(n) elements, for all  $n \in \mathbb{N}$ .

sets given (only) as in the following description:

# unions over $\mathbb{N}$ of finite sets with a width function,

namely via  $B = \bigcup_{n \in \mathbb{N}} B_n$  from (2.3), and vice versa, namely via  $k : [0, 1] \to \mathbb{R}$  as in (2.4). The generalisations of bounded variation from Remark 3.23 have a similar property, as evidenced by the final part of the proof of Theorem 3.20.

Motivated by Remark 2.5, we introduce our alternative (equivalent over ZF) definition of countability for the RM-study of BV-functions.

DEFINITION 2.6. A set  $B \subset \mathbb{R}$  is *height–width countable* if there is a height function  $H : \mathbb{R} \to \mathbb{N}$  and width function  $g : \mathbb{N} \to \mathbb{N}$ , i.e., for all  $n \in \mathbb{N}$ , the set  $B_n := \{x \in B : H(x) < n\}$  is finite with size bound g(n).

Finally, the following technical remark makes the claims in Remarks 2.2 and 2.5 more precise in terms of logical systems.

REMARK 2.7. As discussed above, the sets  $A_n$  from (2.1) and  $B_n$  from (2.3) are finite, while the unions  $A = \bigcup_{n \in \mathbb{N}} A_n$  and  $B = \bigcup_{n \in \mathbb{N}} B_n$  are countable. Hence, working in ZF (or even  $\mathbb{Z}_2^{\Omega}$  from Section 1.3.2), the following objects can be constructed:

- for  $n \in \mathbb{N}$ , an injection  $Y_n$  from  $A_n$  to some  $\{0, 1, \dots, k\}$  with  $k \in \mathbb{N}$ ,
- for  $m \in \mathbb{N}$ , an RM-code  $C_m$  (see [89, II.5.6]) for the closed sets  $A_m$  or  $B_m$ .

However, it is shown in [81, 82] that neither  $Y_n$  nor  $C_n$  are computable (in the sense of Kleene S1–S9; see [58]) in terms of any  $S_m^2$  and the other data. As a result, even  $Z_2^{\omega}$  cannot prove the general existence of  $Y_n$  and  $C_n$  as in the previous items. By contrast, the system ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup> (and even fragments) suffice to show that Definitions 2.3, 2.6, and 2.4 apply to A,  $A_n$  from (2.1) and B,  $B_n$  from (2.3).

In conclusion, we have introduced 'new'—but equivalent over ZF and the weaker  $Z_2^{\Omega}$ —definitions of finite and countable set with the following properties.

- Our 'new' definitions capture the notion of finite and countable set as it occurs 'in the wild', namely in the study of BV or regulated functions. This holds over relatively weak systems by Remark 2.7.
- One finds our 'new' definitions, in particular the notion of 'height', in the literature (see [40, 49, 59, 75, 97]).
- These 'new' definitions shall be seen to yield many equivalences in the RM of the uncountability of  $\mathbb{R}$  (Sections 3.3 and 3.4).

We believe that the previous items justify our adoption of our 'new' definitions of finite and countable set. Moreover, Remark 2.1 creates some historical precedent based on second-order RM and constructive mathematics.

## §3. Main results: regulated and BV-functions.

**3.1. Introduction.** In this section, we establish the equivalences sketched in Section 1.1 pertaining to the uncountability of  $\mathbb{R}$  and properties of regulated functions (Section 3.3) and *BV*-functions (Section 3.4). In Section 3.2, we establish some basic properties of *BV* and regulated functions in weak systems.

As noted in Section 1.1, we shall show that the uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub> is *robust*, i.e., equivalent to small perturbations of itself [60, p. 432]. Striking examples of this claimed robustness may be found in Theorem 3.7, where the perturbations are given by considering either one point of continuity, a dense set of such points, or various uncountability criteria for  $C_f$ , the set of continuity points.

Finally, the content of Section 3.5 is explained in the following remark.

REMARK 3.1 (When more is less). As noted in Section 1.1, NIN<sub>alt</sub> is equivalent to well-known theorems from analysis restricted to regulated functions, with similar results for BV-functions. These (restricted) theorems thus imply NIN and are not provable in  $Z_2^{\omega} + QF-AC^{0,1}$  as a result (see [69]). We show in Section 3.5 that adding the extra condition 'Baire 1' to these theorems makes them provable from (essentially) arithmetical comprehension. While regulated and BV-functions are Baire 1, say over ZF or  $Z_2^{\Omega}$ , there is no contradiction here as the statement

# a BV-function on the unit interval is Baire 1

already implies NIN by [72, Theorem 2.34]. We stress that 'Baire 1' is special in this regard: other restrictions, e.g., involving semi-continuity or Baire 2, yield theorems that are still equivalent to  $NIN_{alt}$ , as shown in Section 3.3.3. We have no explanation for this phenomenon.

**3.2. Preliminary results.** We collect some preliminary results pertaining to regulated and BV-functions, and NIN<sub>alt</sub> from Section 1.1.

First of all, to allow for a smooth treatment of finite sets, we shall adopt the following principle that collects the most basic properties of finite sets.

PRINCIPLE 3.2 (FIN).

- Finite union theorem: for a sequence of finite sets  $(X_n)_{n \in \mathbb{N}}$  and any  $k \in \mathbb{N}$ ,  $\bigcup_{n \leq k} X_n$  is finite.
- For any finite  $X \subset \mathbb{R}$ , there is a finite sequence of reals  $(x_0, ..., x_k)$  that includes all elements of X.
- Finite Axiom of Choice: for  $Y^2$ ,  $k^0$  with  $(\forall n \le k)(\exists f \in 2^{\mathbb{N}})(Y(f, n) = 0)$ , there is a finite sequence  $(f_0, ..., f_k)$  in  $2^{\mathbb{N}}$  with  $(\forall n \le k)(Y(f_n, n) = 0)$ .

One can readily derive FIN from a sufficiently general fragment of the induction axiom; the RM of the latter is well-known (see, e.g., [89, X.4.4]) and the RM of (fragments of) FIN is therefore a matter of future research. We note that in [68], we could derive (fragments of) FIN from the principles under study, like the fact that (height) countable sets of reals can be enumerated. Hence, we could mostly avoid the use of fragments of FIN in the base theory in [68], which does not seem possible for this paper.

Secondly, we need some some basic properties of BV and regulated functions, all of which have been established in [68] already.

Theorem 3.3 (ACA<sub>0</sub><sup> $\omega$ </sup>).

- Assuming FIN, any BV-function  $f : [0,1] \rightarrow \mathbb{R}$  is regulated.
- Any monotone function  $f : [0, 1] \rightarrow \mathbb{R}$  has bounded variation.
- For any monotone function  $f : [0, 1] \to \mathbb{R}$ , there is a sequence  $(x_n)_{n \in \mathbb{N}}$  that enumerates all  $x \in [0, 1]$  such that f is discontinuous at x.

- For regulated  $f : [0, 1] \to \mathbb{R}$  and  $x \in [0, 1]$ , f is sequentially continuous at x if and only if f is epsilon-delta continuous at x.
- For finite  $X \subset [0, 1]$ , the function  $\mathbb{1}_X$  has bounded variation.

PROOF. Proofs may be found in [68, Section 3.3].

The fourth item of Theorem 3.3 is particularly interesting as the local equivalence between sequential and epsilon-delta continuity for general  $\mathbb{R} \to \mathbb{R}$ -functions is not provable in ZF, while  $\mathsf{RCA}_0^{\omega} + \mathsf{QF-AC}^{0,1}$  suffices, as discussed in Remark 1.14.

Thirdly, we discuss some 'obvious' equivalences for NIN and NBI.

REMARK 3.4. Now, NIN and NBI are formulated for mappings from [0, 1] to  $\mathbb{N}$ , but we can equivalently replace the unit interval by, e.g.,  $\mathbb{R}$ ,  $2^{\mathbb{N}}$ , and  $\mathbb{N}^{\mathbb{N}}$ , as shown in [79, Section 2.1]. An important observation in this context, and readily formalised in ACA<sub>0</sub><sup> $\omega$ </sup>, is that the (rescaled) tangent function provides a bijection from any open interval to  $\mathbb{R}$ ; the inverse of tangent, called *arctangent*, yields a bijection in the other direction (also with rescaling). Moreover, using these bijections, one readily shows that NIN<sub>alt</sub> is equivalent to the following:

• there is no height function from  $\mathbb{R}$  to  $\mathbb{N}$ .

Similarly, if we can show that there is no height function from some fixed open interval to  $\mathbb{N}$ , then NIN<sub>alt</sub> follows. We will tacitly make use of this fact in the proof of Theorems 3.6 and 3.7.

Fourth, while we choose to use (at least) the system  $ACA_0^{\omega}$  as our base theory, one can replace the latter by  $RCA_0^{\omega}$  using the following trick.

REMARK 3.5 (Excluded middle trick). The law of excluded middle as in  $(\exists^2) \lor \neg(\exists^2)$  is quite useful as follows: suppose we are proving  $T \to \text{NIN}_{\text{alt}}$  over  $\text{RCA}_0^{\omega}$ . Now, in case  $\neg(\exists^2)$ , all functions on  $\mathbb{R}$  are continuous by [48, Proposition 3.12] and  $\text{NIN}_{\text{alt}}$  then trivially<sup>8</sup> holds. Hence, what remains is to establish  $T \to \text{NIN}_{\text{alt}}$  in case we have  $(\exists^2)$ . However, the latter axiom, e.g., implies ACA<sub>0</sub> and can uniformly convert reals to their binary representations. In this way, finding a proof in  $\text{RCA}_0^{\omega} + (\exists^2)$  is 'much easier' than finding a proof in  $\text{RCA}_0^{\omega}$ . In a nutshell, we may without loss of generality assume  $(\exists^2)$  when proving theorems that are trivial (or readily proved) when all functions (on  $\mathbb{R}$  or  $\mathbb{N}^{\mathbb{N}}$ ) are continuous, like  $\text{NIN}_{\text{alt}}$ . Moreover, we can replace  $2^{\mathbb{N}}$  by [0, 1] at will, which is convenient sometimes.

While the previous trick is useful, it should be used sparingly: the axiom  $(\exists^2)$  is required to guarantee that basic sets like the unit interval are sets in our sense (Definition 1.7) or that finite sets (Definition 1.2) are well-behaved. For this reason, we only mention Remark 3.5 in passing and shall generally work over ACA<sub>0</sub><sup> $\omega$ </sup>.

**3.3. Regulated functions and the uncountability of**  $\mathbb{R}$ . We establish the equivalences sketched in Section 1.1 pertaining to the uncountability of  $\mathbb{R}$  and properties of regulated functions.

•

1620

 $\dashv$ 

<sup>&</sup>lt;sup>8</sup>In case  $H : \mathbb{R} \to \mathbb{N}$  is continuous on  $\mathbb{R}$ , the set  $A_n := \{x \in A : H(x) < n\}$  for A = [0, 1] in Definition 2.3 cannot be finite for any  $n \in \mathbb{N}$  for which it is non-empty.

3.3.1. Volterra's early work. In this section, we connect the uncountability of  $\mathbb{R}$  to Volterra's early results from Section 1.2. In particular, we establish the following theorem where the final two items exhibit some nice robustness properties of NIN<sub>alt</sub> and Volterra's results, as promised in Section 1.1.

THEOREM 3.6 (ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup> + FIN). The following are equivalent.

- (a) The uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub>.
- (b) Volterra's theorem for regulated functions: there do not exist two regulated functions defined on the unit interval for which the continuity points of one are the discontinuity points of the other, and vice versa.
- (c) Volterra's corollary for regulated functions: there is no regulated function that is continuous on  $\mathbb{Q} \cap [0, 1]$  and discontinuous on  $[0, 1] \setminus \mathbb{Q}$ .
- (d) Generalised Volterra's corollary (Theorem 1.5) for regulated functions and height countable D (or: countable D, or: strongly countable D).
- (e) For a sequence  $(X_n)_{n \in \mathbb{N}}$  of finite sets in [0, 1], the set  $[0, 1] \setminus \bigcup_{n \in \mathbb{N}} X_n$  is dense (or: not height countable, or: not countable, or: not strongly countable).

**PROOF.** First of all, Volterra's theorem implies Volterra's corollary (both restricted to regulated functions), as Thomae's function T from (1.1) is readily defined using  $\exists^2$ , while the latter also shows that T is regulated and continuous exactly on  $\mathbb{R} \setminus \mathbb{Q}$ .

Secondly, we now derive NIN<sub>alt</sub> from Volterra's corollary as in item (c). To this end, let  $(X_n)_{n \in \mathbb{N}}$  be a sequence of finite sets such that  $[0, 1] = \bigcup_{n \in \mathbb{N}} X_n$ . Now use  $\mu^2$  to define the following function:

$$g(x) := \begin{cases} 0, & x \in \mathbb{Q}, \\ \frac{1}{2^{n+1}}, & x \in \mathbb{R} \setminus \mathbb{Q} \land x \in X_n \text{ and } n \text{ is the least such number.} \end{cases}$$
(3.1)

We have 0 = g(0+) = g(0-) = g(x+) = g(x-) for any  $x \in (0,1)$ , i.e., g is regulated. To establish this fact in our base theory, note that  $\bigcup_{k \le n} X_k$  is finite for any  $n \in \mathbb{N}$  and can be enumerated, both thanks to FIN. As a result, g is continuous at any  $x \in \mathbb{Q} \cap [0, 1]$  and discontinuous at any  $y \in [0, 1] \setminus \mathbb{Q}$ . This contradicts Volterra's corollary (for regulated functions), and NIN<sub>alt</sub> follows.

Thirdly, we derive Volterra's corollary (for regulated functions) from NIN<sub>alt</sub>, by contraposition. To this end, let f be regulated, continuous on  $[0, 1] \cap \mathbb{Q}$ , and discontinuous on  $[0, 1] \setminus \mathbb{Q}$ . Now consider the following set

$$X_n := \left\{ x \in (0,1) : |f(x+) - f(x)| > \frac{1}{2^n} \lor |f(x-) - f(x)| > \frac{1}{2^n} \right\},$$
(3.2)

where we note that, e.g., the right limit f(x+) for  $x \in (0, 1)$  equals  $\lim_{k\to\infty} f(x + \frac{1}{2^k})$ ; the latter limit is arithmetical and hence  $\mu^2$  readily obtains it. Hence, the set  $X_n$  from (3.2) can be defined in ACA<sub>0</sub><sup>o</sup>. To show that  $X_n$  is finite, suppose not and apply QF-AC<sup>0,1</sup> to find a sequence of reals in  $X_n$ . By the Bolzano–Weierstrass theorem from [89, III.2], this sequence has a convergent sub-sequence, say with limit  $c \in [0, 1]$ ; then either f(c -) or f(c+) does not exist (using the usual epsilon-delta definition), a contradiction. Hence,  $X_n$  is finite and by the assumptions on f, we have  $D_f = \bigcup_{n \in \mathbb{N}} X_n = [0, 1] \setminus \mathbb{Q}$ . Then  $[0, 1] = D_f \cup \mathbb{Q} = (\bigcup_{n \in \mathbb{N}} X_n) \cup \mathbb{Q}$  shows that the unit interval is a union over  $\mathbb{N}$  of finite sets, i.e.,  $\neg \text{NIN}_{\text{alt}}$  follows. One derives item (b) from NIN<sub>alt</sub> in the same way; indeed:  $[0, 1] = D_f \cup D_g$  in case item (b) is false

for regulated  $f, g: [0, 1] \to \mathbb{R}$ , showing that the unit interval is the union over  $\mathbb{N}$  of finite sets, yielding  $\neg NIN_{alt}$ .

Fourth, we only need to show that NIN<sub>alt</sub> implies item (d), as  $\mathbb{Q}$  is trivially (height) countable and dense. Hence, let *f* be regulated, continuous on  $[0, 1] \cap D$ , and discontinuous on  $[0, 1] \setminus D$ , where *D* is height countable and dense. In particular, assume  $D = \bigcup_{n \in \mathbb{N}} D_n$  where  $D_n$  is finite for  $n \in \mathbb{N}$ . Now consider  $X_n$  as in (3.2) from the above and note that  $[0, 1] \setminus D = \bigcup_{n \in \mathbb{N}} X_n$ . Hence,  $[0, 1] = \bigcup_{n \in \mathbb{N}} Y_n$  where  $Y_n = X_n \cup D_n$  is finite (as the components are), i.e.,  $\neg NIN_{alt}$  follows.

Finally, we only need to show that NIN<sub>alt</sub> implies the final item (e). For the terms in brackets in the latter, this is trivial as (strongly) countable sets are height countable. For the density claim, let  $(X_n)_{n \in \mathbb{N}}$  be a sequence of finite sets and suppose  $x_0 \in [0, 1]$  and  $N_0 \in \mathbb{N}$  are such that  $B(x_0, \frac{1}{2^{N_0}}) \cap ([0, 1] \setminus \bigcup_{n \in \mathbb{N}} X_n)$  is empty. Hence,  $B(x_0, \frac{1}{2^{N_0}}) \subset \bigcup_{n \in \mathbb{N}} X_n$  and define the finite sets  $Y_n := X_n \cap B(x_0, \frac{1}{2^{N_0}})$  using  $\exists^2$ . This implies  $B(x_0, \frac{1}{2^{N_0}}) = \bigcup_{n \in \mathbb{N}} Y_n$ , which contradicts NIN<sub>alt</sub>, modulo the rescaling discussed in Remark 3.4.

The final item of the theorem essentially expresses the Baire category theorem restricted to the complement of finite sets, which are automatically open and dense.

3.3.2. Continuity and Riemann integration. We connect the uncountability of  $\mathbb{R}$  to properties of regulated functions like continuity and Riemann integration.

First of all, we shall need the set of (dis)continuity points of regulated  $f : [0, 1] \rightarrow \mathbb{R}$ , definable via  $\exists^2$  as follows:

$$C_f := \{x \in (0,1) : f(x) = f(x+) = f(x-)\} \text{ and } D_f = [0,1] \setminus C_f.$$

These sets occupy a central spot in the study of regulated functions. We have the following theorem, where most items exhibit some kind of robustness.

THEOREM 3.7 (ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup> + FIN). The following are equivalent.

- (i) The uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub>.
- (ii) For any regulated  $f : [0, 1] \to \mathbb{R}$ , there is  $x \in [0, 1]$  where f is continuous (or: *quasi-continuous*, or: *lower semi-continuous*).
- (iii) Any regulated  $f : [0, 1] \to \mathbb{R}$  is pointwise discontinuous, i.e., the set  $C_f$  is dense in the unit interval.
- (iv) For regulated  $f : [0,1] \to \mathbb{R}$ , the set  $C_f$  is not height countable (or: not countable, or: not strongly countable, or: not enumerable).
- (v) For regulated  $f : [0, 1] \to [0, 1]$  such that the Riemann integral  $\int_0^1 f(x) dx$  exists and is 0, there is  $x \in [0, 1]$  with f(x) = 0 Bourbaki ([10, p. 61]).
- (vi) For regulated  $f : [0, 1] \rightarrow [0, 1]$  such that the Riemann integral  $\int_0^1 f(x) dx$  exists and equals 0, the set  $\{x \in [0, 1] : f(x) = 0\}$  is dense ([10, p. 61]).
- (vii) Blumberg's theorem [8] restricted to regulated functions on [0, 1].
- (viii) Measure theoretic Blumberg's theorem [16]: for regulated  $f : [0, 1] \rightarrow \mathbb{R}$ , there is a dense and uncountable (or: not strongly countable, or: not height countable) subset  $D \subset [0, 1]$  such that  $f_{\uparrow D}$  is pointwise discontinuous.
- (ix) For regulated  $f : [0, 1] \to (0, 1]$ , there exist  $N \in \mathbb{N}$ ,  $x \in [0, 1]$  such that  $(\forall y \in B(x, \frac{1}{2^N}))(f(y) \ge \frac{1}{2^N})$ .

- (x) For regulated  $f : [0, 1] \to (0, 1]$ , there exist a dense set D such that  $f_{\uparrow D}$  is locally bounded away from zero<sup>9</sup> (on D).
- (xi) (FTC) For regulated  $f : [0, 1] \to \mathbb{R}$  such that  $F(x) := \lambda x$ .  $\int_0^x f(t) dt$  exists, there is  $x_0 \in (0, 1)$  where F(x) is differentiable with derivative  $f(x_0)$ .
- (xii) For any regulated  $f : [0, 1] \to \mathbb{R}$ , there is a Darboux point.
- (xiii) For any regulated  $f : [0, 1] \to \mathbb{R}$ , its Darboux points are dense.
- (xiv) For any regulated  $f : [0, 1] \to \mathbb{R}$  with only removable discontinuities, there is  $x \in [0, 1]$  which is not a strict<sup>10</sup> local maximum.

**PROOF.** First of all, we prove item (ii) from NIN<sub>alt</sub>; we may use Volterra's corollary as in Theorem 3.6. Fix regulated  $f : [0, 1] \rightarrow \mathbb{R}$  and consider this case distinction:

- If there is  $q \in \mathbb{Q} \cap [0, 1]$  with f(q+) = f(q-) = f(q), item (ii) follows.
- If there is no such rational, then Volterra's corollary guarantees there is  $x \in [0, 1] \setminus \mathbb{Q}$  such that f is continuous at x.

In each case, there is a point of continuity for f, i.e., item (ii) follows. To prove that item (ii) implies NIN<sub>alt</sub>, let  $X := \bigcup_{n \in \mathbb{N}} X_n$  be the union of finite sets  $X_n \subset [0, 1]$  and define h as in (2.2). As for g from (3.1) in Theorem 3.6, h is regulated. Item (ii) then provides a point of continuity  $y \in [0, 1]$  of h, which by definition must be such that  $y \notin X$ . The same holds for quasi- and lower semi-continuity.

The implication (iii) $\rightarrow$ (ii) is immediate (as the empty set is not dense in [0, 1]). To prove (i) $\rightarrow$ (iii), let f be regulated and such that  $C_f$  is not dense. To derive  $\neg$ NIN<sub>alt</sub>, consider  $X_n$  as in (3.2), which is finite for all  $n \in \mathbb{N}$  (as is proved using QF-AC<sup>0,1</sup> and the Bolzano–Weierstrass theorem). Since  $C_f$  is not dense in [0, 1], there is  $y \in [0, 1]$  and  $N \in \mathbb{N}$  such that  $B(y, \frac{1}{2^N}) \cap C_f = \emptyset$ . By definition, the set  $D_f = \bigcup_{n \in \mathbb{N}} X_n$  collects all points where f is discontinuous. Hence,  $[0, 1] \setminus C_f = D_f$ , yielding  $B(y, \frac{1}{2^N}) \subset D_f$ . Now define  $Y_n = X_n \cap B(y, \frac{1}{2^N})$ , which is finite since  $X_n$ is finite. Hence,  $\bigcup_{n \in \mathbb{N}} Y_n = B(y, \frac{1}{2^N})$ , i.e., an interval can be expressed as the union over  $\mathbb{N}$  of finite sets, which readily yields  $\neg$ NIN<sub>alt</sub> after rescaling as in Remark 3.4.

Regarding item (iv), it suffices to derive the latter from NIN<sub>alt</sub>, which is immediate as  $D_f$  is height countable. Indeed, if  $C_f$  is also height countable, then  $[0, 1] = C_f \cup D_f$  is height countable, contradicting NIN<sub>alt</sub>. In case  $A \subset [0, 1]$  is countable, then any  $Y : [0, 1] \rightarrow \mathbb{N}$  injective on A is also a height function, i.e., A is also height countable. The same holds for strongly countable and enumerable sets.

Regarding items (v) and (vi), the latter immediately follow from items (ii) and (iii). Indeed, in case f(x) > 0 for  $x \in C_f$ , then by continuity there are  $k, N \in \mathbb{N}$  such that  $f(y) > \frac{1}{2^k}$  for  $y \in B(x, \frac{1}{2^{N+1}})$ , implying  $\int_0^1 f(x)dx > \frac{1}{2^k2^N} > 0$ . Now assume item (v), let  $(X_n)_{n\in\mathbb{N}}$  be a sequence of finite sets, and let h be as in (2.2). The latter is Riemann integrable with  $\int_0^1 h(x)dx = 0$ , which one shows via the usual 'epsilon-delta' definition and FIN. Any  $y \in [0, 1]$  such that h(y) = 0, also satisfies  $y \notin \bigcup_{n\in\mathbb{N}} X_n$ , i.e., NIN<sub>alt</sub> follows.

<sup>&</sup>lt;sup>9</sup>In symbols:  $(\forall x \in D)(\exists N \in \mathbb{N})(\forall y \in B(x, \frac{1}{2^N}) \cap D)(f(y) \ge \frac{1}{2^N})$ , where we stress the underlined part as it implements the claimed restriction to D.

<sup>&</sup>lt;sup>10</sup>A point  $x \in [0, 1]$  is a strict local maximum of  $f : [0, 1] \to \mathbb{R}$  in case  $(\exists N \in \mathbb{N})(\forall y \in B(x, \frac{1}{2^N}))(y \neq x \to f(y) < f(x))$ .

Next, we clearly have (iii) $\rightarrow$ (vii) and (iv) $\rightarrow$ (viii) since  $D = C_f$  is as required. To prove (vii) $\rightarrow$  NIN<sub>alt</sub>, let  $(X_n)_{n\in\mathbb{N}}$  be a sequence of finite sets in [0, 1] and consider the regulated function h from (2.2). Let D be the dense set provided by item (vii) and consider  $y \in D$ . In case  $h(y) \neq 0$ , say  $h(y) > \frac{1}{2^{k_0}}$ , use FIN to enumerate  $\bigcup_{n \leq k_0+1} X_n$ . Hence, we can find  $N \in \mathbb{N}$  such that for  $z \in B(x, \frac{1}{2^N})$ ,  $h(z) < \frac{1}{2^{k_0+1}}$ , i.e.,  $h_{\uparrow D}$  is not continuous (on D). This contradiction implies that h(y) = 0, meaning  $y \in [0, 1] \setminus$  $\bigcup_{n \in \mathbb{N}} X_n$ , i.e., NIN<sub>alt</sub> follows. A similar proof works for item (viii) by considering a point from the dense set of continuity points of  $f_{\uparrow D}$ .

Next, (ii) (resp. (iii)) clearly implies (ix) (resp. (x)). To show that (ix) and (x) imply NIN<sub>alt</sub>, one proceeds as in the previous paragraphs. Similarly, (ii) (resp. (iii)) implies (xii) (resp. (xiii)), as any continuity point is a Darboux point, by definition. To show that (xii) implies NIN<sub>alt</sub>, one considers the function *h* as in (2.2) and notes that a Darboux point of *h* is not in  $\bigcup_{n \in \mathbb{N}} X_n$ .

For item (xi), the usual epsilon-delta proof establishes that  $F(x) := \int_0^x f(t)dt$  is continuous and that F'(y) = f(y) in case f is continuous at y, i.e., item (ii) implies item (xi). As noted above, h as in (2.2) satisfies  $H(x) := \int_0^x h(x)dt = 0$  for any  $x \in [0, 1]$ , i.e., any  $y \in (0, 1)$  such that H'(y) = 0 = h(y) is such that  $y \notin \bigcup_{n \in \mathbb{N}} X_n$ , yielding NIN<sub>alt</sub> as required.

Finally, assume item (xiv), let  $(X_n)_{n \in \mathbb{N}}$  be a sequence of finite sets, and note that h as in (2.2) is regulated with only removable discontinuities. Now, the set  $X = \bigcup_{n \in \mathbb{N}} X_n$ consists of the local strict maxima of h, i.e., item (xiv) yields NIN<sub>alt</sub>. For the reversal,  $\exists^2$  computes a functional M such that M(g, a, b) is a maximum of  $g \in C([0, 1])$  on  $[a, b] \subset [0, 1]$  (see [48, Section 3]), i.e.,  $(\forall y \in [a, b])(g(y) \leq g(M(g, a, b)))$ . Using the functional M, one readily shows that 'x is a strict local maximum of g' is decidable<sup>11</sup> given  $\exists^2$ , for g continuous on [0, 1]. Now let  $f : [0, 1] \to \mathbb{R}$  be regulated and with only removable discontinuities. Use  $\exists^2$  to define  $\tilde{f}: [0,1] \to \mathbb{R}$  as follows:  $\tilde{f}(x) := f(x+)$  for  $x \in [0,1)$  and  $\tilde{f}(1) = f(1-)$ . By definition,  $\tilde{f}$  is continuous on [0, 1], and  $\exists^2$  computes a (continuous) modulus of continuity, which follows in the same way as for Baire space (see, e.g., [47, Section 4]). In case f is discontinuous at  $x \in [0, 1]$ , the latter point is a strict local maximum of f if and only if f(x) > 1f(x+) (or f(x) > f(x-) in case x = 1). Note that  $\mu^2$  (together with a modulus of continuity for  $\tilde{f}$ ) readily yields  $N_{f,x} \in \mathbb{N}$  such that  $(\forall y \in B(x, \frac{1}{2^{N_{f,x}}}))(f(y) < 0$ f(x), in case x is a strict local maximum of f. In case f is continuous at  $x \in [0, 1]$ , we can use  $\exists^2$  to decide whether x is a local strict maximum of  $\tilde{f}$ . By Footnote 11,  $\mu^2$  again yields  $N_{f,x} \in \mathbb{N}$  such that  $(\forall y \in B(x, \frac{1}{2N}))(\tilde{f}(y) < \tilde{f}(x)))$ , in case x is a strict local maximum of  $\tilde{f}$ . Now consider the following set:

 $A_n := \{x \in [0, 1] : x \text{ is a strict local maximum of } \tilde{f} \text{ or } f \text{ with } n \ge N_{f, x}\}.$ 

<sup>11</sup>If  $g \in C([0, 1])$ , then  $x \in [0, 1]$  is a strict local maximum iff for some  $\varepsilon \in \mathbb{Q}^+$ :

- g(y) < g(x) whenever  $|x y| < \varepsilon$  for any  $q \in [0, 1] \cap \mathbb{Q}$ , and
- $\sup_{y \in [a,b]} g(y) < g(x)$  whenever  $x \notin [a,b]$ ,  $a, b \in \mathbb{Q}$  and  $[a,b] \subset [x \varepsilon, x + \varepsilon]$ .

Note that  $\mu^2$  readily yields  $N \in \mathbb{N}$  such that  $(\forall y \in B(x, \frac{1}{2^N}))(g(y) < g(x))$ .

Then  $A_n$  is finite as strict local maxima cannot be 'too close'. Hence, NIN<sub>alt</sub> yields  $y \in [0, 1] \setminus \bigcup_{n \in \mathbb{N}} A_n$ , which is not a local maximum of f by definition, i.e., item (xiv) follows, and we are done.

We may view item (i) as an extremely basic version of the connectedness of [0, 1], as defined by Jordan in [43, pp. 24–28]. Similarly, item (xi) is an extremely basic version of the fundamental theorem of calculus (FTC) and item (ii) is an extremely basic version of the Lebesgue criterion for Riemann integrability. Moreover, it seems necessary to formulate items (v), (vi), and (xi) with the extra condition that the functions at hand be Riemann integrable. As an exercise, the reader should prove that  $h : [0, 1] \rightarrow \mathbb{R}$  as in (2.2) is *effectively* Riemann integrable, i.e., there is a functional that outputs the ' $\delta > 0$ ' on input the ' $\varepsilon > 0$ ' as in the usual epsilon-delta definition of Riemann integrability.

Moreover, the notion of 'left and right limits' gives rise to the notion of 'left and right derivatives'; following item (xi), the left (resp. right) limit of regulated  $f : [0, 1] \rightarrow \mathbb{R}$  equals the left (resp. right) derivative of F at every  $x_0 \in (0, 1)$  via a completely elementary proof (say in ACA<sub>0</sub><sup> $\omega$ </sup>). We could also formulate item (ii) with *approximate continuity* (see, e.g., [17, II.5]) in the conclusion, but this notion seems to involve a lot of measure theory.

*3.3.3. Restrictions of regulated functions.* In this section, we show that the above equivalences for NIN<sub>alt</sub> remain valid if we impose certain natural restrictions.

First of all, the above results show that we have to be careful with intuitive statements like *regulated functions are 'close to continuous'*. Indeed, by Theorem 3.7,  $Z_2^{\omega} + QF-AC^{0,1}$  is consistent with the existence of regulated functions that are discontinuous everywhere. Similarly, NIN follows from the statement *a regulated function is Baire 1* by [72, Theorem 2.34]. Hence,  $Z_2^{\omega} + QF-AC^{0,1}$  cannot prove the latter basic fact and the restriction in item (ii) in Theorem 3.8 is therefore non-trivial. Similar results hold for items items (iii)–(iv) in Theorem 3.8 following [72, Section 2.8].

THEOREM 3.8 (ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup> + FIN). The following are equivalent.

- (i) The uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub>.
- (ii) For any regulated and Baire 2 function  $f : [0, 1] \rightarrow \mathbb{R}$ , there is  $x \in [0, 1]$  where *f* is continuous.
- (iii) For any regulated and Baire  $1^*$  function  $f : [0, 1] \to \mathbb{R}$ , there is  $x \in [0, 1]$  where *f* is continuous.
- (iv) For any regulated and usco (or: lsco) function  $f : [0, 1] \rightarrow \mathbb{R}$ , there is  $x \in [0, 1]$  where f is continuous.

**PROOF.** In light of Theorem 3.7, we only need to show that items (ii)–(iv) imply NIN<sub>alt</sub>. The function  $h : [0, 1] \to \mathbb{R}$  from (2.2) is central in the proof of the former theorem. For item (ii), h is Baire 2 as follows: define  $h_n(x)$  as h(x) in case  $x \in \bigcup_{m \le n} X_n$ , and 0 otherwise. By definition, h is the pointwise limit of  $(h_n)_{n \in \mathbb{N}}$  and FIN allows us to enumerate  $\bigcup_{m \le n_0} X_n$  for fixed  $n_0 \in \mathbb{N}$ . Hence, for fixed  $n_0 \in \mathbb{N}$ ,  $h_{n_0}$ has at most finitely many points of discontinuity. In particular, there is an obvious sequence of continuous function with pointwise limit  $h_{n_0}$ , i.e., the latter is Baire 1. For item (iv), the function h is also usco as follows: in case  $h(x_0) = 0$  for  $x_0 \in [0, 1]$ ,

the definition of usco is trivially satisfied. In case  $h(x_0) = \frac{1}{2^{n+1}}$ , then there is  $N \in \mathbb{N}$  such that  $(\forall y \in B(x, \frac{1}{2^N}))(y \notin \bigcup_{m \leq n-1} X_m)$  as in the proof of Theorem 3.7. In this case, the definition of usco is also satisfied. The function  $\lambda x.(1 - h(x))$  is lsco. For item (iii), define the closed sets  $C_n := \bigcup_{m \leq n} X_m$  and note that the restriction of *h* to  $C_n$  is continuous for each *n*, i.e., *h* is also Baire 1<sup>\*</sup>.

Secondly, we point out one subtlety in the previous proof: it is *only* shown that  $h : [0, 1] \to \mathbb{R}$  from (2.2) is Baire 2. In particular, we cannot<sup>12</sup> construct a double sequence of continuous functions with iterated limit equal to *h*. As it happens, Baire himself notes in [3, p. 69] that Baire 2 functions can be *represented* by such double sequences. We *could* generalise item (ii) in Theorem 3.8 to any higher Baire class and beyond, i.e., the latter theorem constitutes robustness in the flesh.

Moreover, going against our intuitions, we cannot replace 'Baire 2' by 'Baire 1' in Theorem 3.8 as the latter condition renders items (ii)–(iv) provable in  $ACA_0^{\omega} + QF-AC^{0,1}$  by the results in Section 3.5. In particular, while Baire 1\* and usco are subclasses of Baire 1, say in ZF or  $Z_2^{\Omega}$ , these inclusions do not necessarily hold in weaker systems. For instance, it is consistent with  $Z_2^{\omega} + QF-AC^{0,1}$  that there are totally discontinuous usco and regulated functions (see Theorem 3.7). An interesting RM-question would be to calibrate the strength of some of the well-known inclusions, like *a regulated function on the unit interval is Baire 1*.

Thirdly, the *supremum principle* for regulated functions implies NIN by [72, Theorem 2.32] where the former principle states the existence of  $F : \mathbb{Q}^2 \to \mathbb{R}$  such that  $F(p,q) = \sup_{x \in [p,q]} f(x)$  for any  $p,q \in [0,1] \cap \mathbb{Q}$ . Indeed, using the well-known interval-halving technique, a supremum functional for  $h : [0,1] \to \mathbb{N}$  as in (2.2) would allow us to enumerate the associated union  $\bigcup_{n \in \mathbb{N}} X_n$ , i.e., NIN<sub>alt</sub> readily follows. Perhaps surprisingly, the equivalences for NIN<sub>alt</sub> from the previous sections still go through if we restrict to regulated functions with a supremum functional. Regarding item (iv), the Heaviside function is regulated but not symmetrically continuous, where the latter notion goes back to Hausdorff [32].

THEOREM 3.9 (ACA<sub>0</sub><sup> $\omega$ </sup> + QF-AC<sup>0,1</sup> + FIN). The following are equivalent.

- (i) The uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub>.
- (ii) For regulated  $f : [0, 1] \to \mathbb{R}$  with a supremum functional, there is  $x \in [0, 1]$  where f is continuous.
- (iii) For regulated and lsco  $f : [0, 1] \to \mathbb{R}$  with a supremum functional, there is  $x \in [0, 1]$  where f is continuous.
- (iv) For regulated and symmetrically continuous  $f : [0, 1] \rightarrow \mathbb{R}$ , there is  $x \in [0, 1]$  where f is continuous.

**PROOF.** The first item implies the other items by Theorem 3.7. Now assume the second item and let  $Y : [0, 1] \to \mathbb{N}$  be an injection and define  $e(x) := \sum_{n=0}^{Y(x)+1} \frac{x^n}{n!}$ . By definition, we have  $e(x) < e^x$  for  $x \in [0, 1]$ . Using the second item of FIN, we have  $e(x+) = e(x-) = e^x$  for  $x \in (0, 1)$ . Indeed, for small enough neighbourhoods U

<sup>&</sup>lt;sup>12</sup>The results in [72, Section 2.6] establish that  $ACA_0^{\omega} + ATR_0$  plus extra induction can prove numerous theorems about *BV*-functions *if* we assume the latter are also Baire 1. This can be generalised from '*BV*' to 'regulated' and from 'Baire 1' to 'Baire 2 given as an iterated limit of a double sequence of continuous functions'. The technical details are, however, rather involved.

of  $x \in (0, 1)$ , Y is arbitrarily large on  $U \setminus \{x\}$ , while  $e^x$  is uniformly continuous on [0, 1]. Hence,  $\lambda x.e(x)$  is regulated (and lsco) and  $\sup_{x \in [p,q]} e(x) = e^q$  also follows. Since the former function is totally discontinuous, we obtain a contradiction. To show that  $\lambda x.e(x)$  is also symmetrically continuous, note that the second item of FIN implies that |e(x + h) - e(x - h)| is arbitrarily small for small enough  $h \in \mathbb{R}$ .

Finally, there are a number of equivalent definitions of 'Baire 1' on the reals [4, 52, 57], including the following ones by [52, Theorem 2.3] and [55, Section 34, VII].

DEFINITION 3.10.

- Any f: [0, 1] → ℝ is *fragmented* if for any ε > 0 and closed C ⊂ [0, 1], there is non-empty relatively<sup>13</sup> open O ⊂ C such that diam(f(O)) < ε.</li>
- Any  $f : [0, 1] \to \mathbb{R}$  is *B-measurable of class 1* if for every open  $Y \subset \mathbb{R}$ , the set  $f^{-1}(Y)$  is  $\mathbf{F}_{\sigma}$ , i.e., a union over  $\mathbb{N}$  of closed sets.

The *diameter* of a set X of reals is defined as usual, namely  $diam(X) := \sup_{x,y \in X} |x - y|$ , where the latter supremum need not exist for Definition 3.10. We have the following theorem, similar to Theorem 3.8.

THEOREM 3.11 (ACA<sub>0</sub><sup> $\omega$ </sup> + IND<sub>0</sub>). The following are equivalent.

- (a) The uncountability of  $\mathbb{R}$  as in NIN<sub>alt</sub>.
- (b) For fragmented and regulated  $f : [0, 1] \to \mathbb{R}$ , there is a point  $x \in [0, 1]$  where f is continuous.
- (c) For *B*-measurable of class 1 and regulated  $f : [0, 1] \to \mathbb{R}$ , there is  $x \in [0, 1]$  where *f* is continuous.

**PROOF.** In light of Theorem 3.7, it suffices to prove that *h* from (2.2) is fragmented and *B*-measurable of class 1. For the former notion, for fixed  $k \in \mathbb{N}$ , FIN can enumerate the (finitely many)  $x \in [0, 1]$  such that  $h(x) \ge \frac{1}{2^k}$ . Any open set *O* not including these points is such that diam $(h(O)) < \frac{1}{2^k}$ , showing that *h* is fragmented.

For the *B*-measurability (of first class), in case  $(X_n)_{n \in \mathbb{N}}$  is a sequence of finite sets such that  $[0, 1] = \bigcup_{n \in \mathbb{N}} X_n$ , note that for any  $Z \subset \mathbb{R}$ , the set  $h^{-1}(Z)$  is the union of those  $X_n$  such that  $\frac{1}{2^{n+1}} \in Z$ , i.e.,  $\mathbf{F}_{\sigma}$  by definition.

We show in Section 3.5 that most of the above statements that are equivalent to NIN<sub>alt</sub>, become provable in the much weaker system  $ACA_0^{\omega} + QF-AC^{0,1} + FIN$  if we additionally require the functions to be Baire 1 *as in Definition* 1.8.

## 3.4. Bounded variation and the uncountability of $\mathbb{R}$ .

3.4.1. Introduction. In this section, we establish the equivalences sketched in Section 1.1 pertaining to the uncountability of  $\mathbb{R}$  and properties of BV-functions. In particular, we study the following weakening of NIN<sub>alt</sub> involving the notion of height-width countability from Definition 2.6.

**PRINCIPLE 3.12** (NIN'<sub>alt</sub>). *The unit interval is not height–width countable.* 

<sup>&</sup>lt;sup>13</sup>For  $A \subseteq B \subset \mathbb{R}$ , we say that *A* is relatively open (in *B*) if for any  $a \in A$ , there is  $N \in \mathbb{N}$  such that  $B(x, \frac{1}{2N}) \cap B \subset A$ . Note that *B* is always relatively open in itself.

Equivalences for  $NIN'_{alt}$  will involve some (restrictions of) items from Theorems 3.6 and 3.7, but also a number of theorems from analysis that hold for *BV*-functions and not for regulated ones. For the latter, we need some additional definitions, found in Section 3.4.2, while the equivalences are in Section 3.4.3

Finally, we first establish Theorem 3.15, which is interesting because we are unable to derive NIN<sub>alt</sub> from the items listed therein. The exact definitions of HBU and WHBU are below, where the former expresses that the uncountable covering  $\bigcup_{x \in [0,1]} B(x, \Psi(x))$  has a finite sub-covering, i.e., the *Heine-Borel theorem* or *Cousin lemma* [21]. The principle WHBU is the combinatorial essence of the Vitali covering theorem, as studied in [66].

PRINCIPLE 3.13 (HBU [65]). For any  $\Psi : \mathbb{R} \to \mathbb{R}^+$ , there are  $y_0, \ldots, y_k \in [0, 1]$  such that  $\bigcup_{i \le k} B(y_i, \Psi(y_i))$  covers [0, 1].

**PRINCIPLE** 3.14 (WHBU [66]). For any  $\Psi : \mathbb{R} \to \mathbb{R}^+$  and  $\varepsilon >_{\mathbb{R}} 0$ , there are  $y_0, \ldots, y_k \in [0, 1]$  such that  $\bigcup_{i \le k} B(y_i, \Psi(y_i))$  has measure at least  $1 - \varepsilon$ .

We note that WHBU can be formulated without using the Lebesgue measure, as discussed at length in e.g., [66] or [89, X.1]. We conjecture that NIN<sub>alt</sub> is not provable in ACA<sub>0</sub><sup> $\omega$ </sup> + HBU.

THEOREM 3.15 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN). The following theorems imply NIN'<sub>alt</sub>:

- Jordan decomposition theorem for [0, 1],
- the principle HBU restricted to BV-functions,
- the principle WHBU restricted to BV-functions.

**PROOF.** For the first item, let  $(Y_n)_{n \in \mathbb{N}}$  be a sequence of finite sets with width bound  $g \in \mathbb{N}^{\mathbb{N}}$ . The function  $k : [0, 1] \to \mathbb{R}$  from (2.4) has bounded variation, with upper bound 1 by definition, for which we need FIN. By [70, Lemma 7],  $\mu^2$  can enumerate the points of discontinuity of a monotone function, i.e., the Jordan decomposition theorem provides a sequence  $(x_n)_{n \in \mathbb{N}}$  that enumerates the points of discontinuity of a *BV*-function. Using the usual diagonal argument (see, e.g., [89, II.4.9]), we can find a point not in this sequence, yielding NIN'<sub>alt</sub>.

For the remaining items, let  $(Y_n)_{n \in \mathbb{N}}$  again be a sequence of finite sets with width bound  $g \in \mathbb{N}^{\mathbb{N}}$ . Suppose  $[0, 1] = \bigcup_{n \in \mathbb{N}} Y_n$  and define  $\Psi : [0, 1] \to \mathbb{R}^+$  as follows  $\Psi(x) := \frac{1}{2^{n+5}} \frac{1}{g(n)+1}$  where  $x \in Y_n$  and n is the least such number. For  $x_0, \ldots, x_k \in$ [0, 1], the measure of  $\bigcup_{i \leq k} B(x_i, \Psi(x_i))$  is at most 1/2 by construction, contradicting HBU and WHBU. Using FIN, one readily shows that  $\Psi$  is in BV.

The principle HBU is studied in [5, 6] for  $\Psi$  represented by, e.g., second-order Borel codes. This 'coded' version is provable in ATR<sub>0</sub> extended with some induction. By contrast and Theorem 3.15, HBU restricted to *BV*-functions, which are definitely Borel, implies NIN'<sub>alt</sub>, which in turn is not provable in Z<sub>2</sub><sup> $\omega$ </sup>. Thus, the use of codes fundamentally changes the logical strength of HBU. A similar argument can be made for the Jordan decomposition theorem, studied for second-order codes in [64].

3.4.2. Definitions. We introduce some extra definitions needed for the RM-study of BV-functions as in Section 3.4.3.

First of all, we shall study *unordered sums*, which are a device for bestowing meaning upon 'uncountable sums'  $\sum_{x \in I} f(x)$  for any index set I and  $f: I \to \mathbb{R}$ .

A central result is that if  $\sum_{x \in I} f(x)$  somehow exists, it must be a 'normal' series of the form  $\sum_{i \in \mathbb{N}} f(y_i)$ , i.e., f(x) = 0 for all but countably many  $x \in [0, 1]$ ; Tao mentions this theorem in [94, p. xii].

By way of motivation, there is considerable historical and conceptual interest in this topic: Kelley notes in [44, p. 64] that Moore's study of unordered sums in [61] led to the concept of *net* with his student Smith [62]. Unordered sums can be found in (self-proclaimed) basic or applied textbooks [39, 90] and can be used to develop measure theory [44, p. 79]. Moreover, Tukey shows in [96] that topology can be developed using *phalanxes*, which are nets with the same index sets as unordered sums.

Now, unordered sums are just a special kind of *net* and  $a: [0, 1] \to \mathbb{R}$  is therefore written  $(a_x)_{x \in [0,1]}$  in this context to suggest the connection to nets. The associated notation  $\sum_{x \in [0,1]} a_x$  is purely symbolic. We only need the following notions in the below. Let  $fin(\mathbb{R})$  be the set of all finite sequences of reals without repetitions.

DEFINITION 3.16. Let  $a : [0, 1] \to \mathbb{R}$  be any mapping, also denoted  $(a_x)_{x \in [0, 1]}$ .

- We say that  $\sum_{x \in [0,1]} a_x$  is *Cauchy* if there is  $\Phi : \mathbb{R} \to fin(\mathbb{R})$  such that for  $\varepsilon > 0$
- and all  $J \in \operatorname{fin}(\mathbb{R})$  with  $J \cap \Phi(\varepsilon) = \emptyset$ , we have  $|\sum_{x \in J} a_x| < \varepsilon$ . We say that  $\sum_{x \in [0,1]} a_x$  is *bounded* if there is  $N_0 \in \mathbb{N}$  such that for any  $J \in \operatorname{fin}(\mathbb{R})$ ,  $N_0 > |\sum_{x \in J} a_x|$ .

Note that in the first item,  $\Phi$  is called a *Cauchy modulus*. For simplicity, we focus on positive unordered sums, i.e.,  $(a_x)_{x \in [0,1]}$  such that  $a_x \ge 0$  for  $x \in [0,1]$ .

Secondly, there are many spaces between the regulated and BV-functions, as discussed in Remark 3.23. We shall study one particular construct, called Waterman variation, defined as follows.

DEFINITION 3.17. A decreasing sequence of positive reals  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$  is a *Waterman sequence* if  $\lim_{n \to \infty} \lambda_n = 0$  and  $\sum_{n=0}^{\infty} \lambda_n = \infty$ .

DEFINITION 3.18 (Waterman variation). The function  $f : [a, b] \to \mathbb{R}$  has bounded *Waterman variation with sequence*  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$  on [a, b] if there is  $k_0 \in \mathbb{N}$  such that  $k_0 \ge \sum_{i=0}^n \lambda_i |f(x_i) - f(x_{i+1})|$  for any finite collection of pairwise non-overlapping intervals  $(x_i, x_{i+1}) \subset [a, b]$ .

Note that Definition 3.18 is *equivalent* to the 'official' definition of Waterman variation by [1, Proposition 2.18]. In case  $f:[0,1] \to \mathbb{R}$  has bounded Waterman variation (with sequence  $\Lambda$  as in Definition 3.17), we write ' $f \in \Lambda BV$ '.

Thirdly, we make use of the usual definitions of Fourier coefficients and series.

DEFINITION 3.19. The Fourier series S(f)(x) of  $f : [-\pi, \pi] \to \mathbb{R}$  at  $x \in [-\pi, \pi]$  is

$$\frac{a_0}{2} + \sum_{n=1}^{\infty} (a_k \cdot \cos(nx) + b_k \cdot \sin(nx)),$$
(3.3)

with Fourier coefficients  $a_n := \frac{1}{\pi} \int_{-\pi}^{\pi} f(t) \cos(nt) dt$  and  $b_n := \frac{1}{\pi} \int_{-\pi}^{\pi} f(t) \sin(nt) dt$ .

By the proof of Theorem 3.22, the Fourier coefficients and series of  $k : [0, 1] \rightarrow \mathbb{R}$ as in (2.2) all exist. Our study of the Fourier series S(f) as in (3.3) will always assume (at least) that the Fourier coefficients of f exist. Functions of bounded variation are of course Riemann integrable and similarly have Fourier coefficients, but only in sufficiently strong systems that seem to dwarf NIN'<sub>alt</sub>.

3.4.3. Basic equivalences. We establish a number of equivalences for NIN'<sub>alt</sub> and basic properties of BV-functions, some similar to those in Theorems 3.6 and 3.7, some new. We note that QF-AC<sup>0,1</sup> is no longer needed in the base theory.

THEOREM 3.20 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN). The following are equivalent.

- (i) The uncountability of  $\mathbb{R}$  as in NIN<sup>'</sup><sub>alt</sub>.
- (ii) For a positive unordered sum  $\sum_{x \in [0,1]} a_x$  with upper bound (or Cauchy modulus; Definition 3.16), there is  $y \in [0,1]$  such that  $a_y = 0$ .
- (iii) For a positive unordered sum  $\sum_{x \in [0,1]} a_x$  with upper bound (or Cauchy modulus; Definition 3.16), the set  $\{x \in [0,1] : a_x = 0\}$  is dense (or: not height countable, or: not countable, or: not strongly countable).
- (iv)-(xix) Any of items (b)-(d) from Theorem 3.6 or items (ii)-(xiv) from Theorem 3.7 restricted to BV-functions.
  - (xx) For a Waterman sequence  $\Lambda$  and  $f : [0, 1] \to \mathbb{R}$  in  $\Lambda BV$ , there is  $y \in [0, 1]$  where f is continuous.
- (xxi)-(xxxvi) Any of items (b)-(d) from Theorem 3.6 or items (ii)-(xiv) from Theorem 3.7 restricted to  $\Lambda BV$  for any fixed Waterman sequence  $\Lambda$ .

PROOF. The equivalences involving the restrictions from regulated to BV functions follow from the proofs of Theorems 3.6 and 3.7. Indeed, for  $f \in BV$  with variation bounded by 1, the set  $X_n$  from (3.2) has size bounded by  $2^n$  since each element of  $X_n$  contributes at least  $1/2^n$  to the variation. Moreover, rather than  $h : [0, 1] \to \mathbb{R}$  as in (2.2), we use  $k : [0, 1] \to \mathbb{R}$  as in (2.4) which has variation bounded by 1 if  $(Y_n)_{n \in \mathbb{N}}$  is a sequence of sets with width function  $g \in \mathbb{N}^{\mathbb{N}}$ . The properties of  $k : [0, 1] \to \mathbb{R}$  are readily proved using FIN; in particular, the width function g obviates the use of QF-AC<sup>0,1</sup> as in the proofs of Theorems 3.6 and 3.7. Hence, the equivalence between NIN'<sub>alt</sub> and items (iv)–(xix) has been established.

The equivalence between items (i) and (ii) is as follows: assume the latter and let  $(X_n)_{n \in \mathbb{N}}$  and  $g : \mathbb{N} \to \mathbb{N}$  be as in item (i). Define  $(a_x)_{x \in [0,1]}$  as follows:

$$a_x := \begin{cases} 0, & x \notin \cup_{n \in \mathbb{N}} X_n, \\ \frac{1}{2^n} \frac{1}{g(n)+1}, & x \in X_n \text{ and } n \text{ is the least such natural.} \end{cases}$$

Clearly, this unordered sum is Cauchy and has upper bound 1; if  $y \in [0, 1]$  is such that  $a_y = 0$ , then  $y \notin \bigcup_{n \in \mathbb{N}} X_n$ , as required for item (i). Now assume the latter and let  $(a_x)_{x \in [0,1]}$  be an unordered sum that is Cauchy. Now consider the following set:

$$X_n := \{ x \in [0, 1] : a_x > 1/2^n \}.$$
(3.4)

Apply the Cauchy property of  $(a_x)_{x \in [0,1]}$  for  $\varepsilon = 1$ , yielding an upper bound  $N_0 \in \mathbb{N}$  for  $\sum_{x \in K} a_x$  for any  $K \in fin(\mathbb{R})$ . Hence, the finite set  $X_n$  in (3.4) has size at most  $2^n N_0$ . In this way, we have a width function for  $(X_n)_{n \in \mathbb{N}}$ ; any  $y \in [0, 1] \setminus \bigcup_{n \in \mathbb{N}} X_n$  is such that  $a_y = 0$ , as required for item (ii). Item (iii) now follows in the same way as for item (e) in the proof of Theorem 3.6.

For item (xx), assume the latter and note that (3.5) establishes that  $f \in BV$ implies  $f \in \Lambda BV$  for *any* Waterman sequence  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$ :

$$\sum_{i=0}^{n} \lambda_i |f(x_i) - f(x_{i+1})| \le \sum_{i=0}^{n} \lambda_1 |f(x_i) - f(x_{i+1})| \le \lambda_1 V_0^1(f), \quad (3.5)$$

as Waterman sequences are decreasing by definition. Hence, item (i) readily follows from item (xx). Now assume item (i) and recall that for *BV*-functions with variation bounded by 1, the set  $X_n$  from (3.2) can have at most  $2^n$  elements, as each element of  $X_n$  contributes at least  $\frac{1}{2^n}$  to the total variation. Functions with Waterman variation bounded by 1 similarly come with explicit upper bounds on the set  $X_n$ , namely  $|X_n| \leq K_n$  where  $K_n$  is the least  $k \in \mathbb{N}$  such that  $2^n < \sum_{m=0}^k \lambda_m$ . Hence, item (xx) follows and we are done.

Following the proof of Theorem 3.8, we observe that we may restrict to BV-function that are Baire 2 or Baire 1<sup>\*</sup> in the previous theorem. With the gift of hindsight we can even obtain the following corollary. We recall that the space of regulated functions is the union over all Waterman sequences  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$  of the spaces  $\Lambda BV$ , as established in [1, Proposition 2.24].

COROLLARY 3.21 (Some generalisations).

- We may replace ' $f \in \Lambda BV$ ' in items (xx)–(xxxvi) of the theorem by 'regulated function  $f : [0, 1] \to \mathbb{R}$  with  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$  such that  $f \in \Lambda BV$ '.
- Assuming  $ACA_0^{\omega} + QF-AC^{0,1} + FIN$ , the higher item implies the lower one. - For any regulated  $f : [0, 1] \rightarrow \mathbb{R}$ , there is a Waterman sequence  $\Lambda =$ 
  - $(\lambda_n)_{n \in \mathbb{N}}$  such that  $f \in \Lambda BV$ .
  - The uniform finite union theorem.

**PROOF.** For the first item, the last part of the proof of the theorem provides the required upper bound function for applying  $NIN'_{alt}$ .

For the second item, let  $(X_n)_{n \in \mathbb{N}}$  be a sequence of finite sets in [0, 1] and consider the regulated function  $h : [0, 1] \to \mathbb{R}$  as in (2.2). Suppose h is in  $\Lambda BV$  with  $\Lambda = (\lambda_n)_{n \in \mathbb{N}}$  and with upper bound  $N_0 \in \mathbb{N}$  on the Waterman variation. Then  $A_{n+2}$  as in (2.1) is  $\bigcup_{i \leq n} X_i$  for all  $n \in \mathbb{N}$  and  $|A_n| \leq K_n$  where  $K_n$  is the least  $k \in \mathbb{N}$  such that  $2^n N_0 < \sum_{m=0}^k \lambda_m$ . The uniform finite union theorem thus follows.

An equivalence is possible in the second item, but the technical details are considerable. Our above results suggest that the principles equivalent to  $NIN'_{alt}$  also have a certain robustness since we can replace 'one point' properties like item (ii) in Theorem 3.20, by, e.g., 'density' versions like item (iii) in Theorem 3.20. Nonetheless, we believe we cannot replace 'density' by 'full measure'. In particular, we conjecture that 'measure theoretic' statements like

- a *BV*-function is continuous (or differentiable) almost<sup>14</sup> everywhere,
- a height–width countable set  $A \subset [0, 1]$  has measure<sup>14</sup> zero

are *strictly* stronger than  $NIN'_{alt}$ . We do not have a proof of this claim.

Finally, the variation function  $\lambda x. V_a^x(f)$  is defined in the obvious way, namely based on (1.2). This function shares pointwise properties like continuity and differentiability with  $f : [a, b] \to \mathbb{R}$ . For instance, the following equivalence for any  $x \in [0, 1)$  is obtained in [37, Corollary 1.1]:

f is right-continuous at x if and only if  $\lambda x. V_a^x(f)$  is right-continuous at x. (3.6)

<sup>&</sup>lt;sup>14</sup>The definition of ' $A \subset [0, 1]$  has measure zero set' can be written down without using the Lebesgue measure, just like in second-order RM (see [89, X.1]).

Here, 'right-continuous at  $y \in [0, 1)$ ' means g(y) = g(y+). Now, although the variation function may not exist for *BV*-functions, say in  $\mathsf{RCA}_0^{\omega}$ , the right-hand side of (3.6) makes sense using the well-known 'virtual' or 'comparative' meaning of suprema from second-order RM (see [89, X.1]). Perhaps surprisingly, working over  $\mathsf{ACA}_0^{\omega} + \mathsf{FIN} + \neg \mathsf{NIN}'_{\mathsf{alt}}$ , the function  $k : [0, 1] \rightarrow \mathbb{R}$  from (2.4) satisfies

the function 
$$\lambda x. V_0^x(k)$$
 is right-continuous for  $x \in [0, 1)$ ,

which is to be interpreted in the aforementioned virtual sense. Thus, one readily proves that the following are equivalent, where E(f, x) is (3.6).

- The uncountability of  $\mathbb{R}$  as in NIN'<sub>alt</sub>.
- For  $f : [0, 1] \to \mathbb{R}$  in BV, there is  $y \in [0, 1]$  where f is continuous.
- For  $f : [0,1] \to \mathbb{R}$  in BV, there is  $y \in [0,1]$  where E(f, y) holds.
- For  $f : [0,1] \to \mathbb{R}$  in *BV* such that  $\lambda x. V_0^x(f)$  is right-continuous on [0, 1), there is  $y \in [0,1]$  where f is right-continuous.

To be absolutely clear, we think this topic should *not* be pursued further: mistakes are (too) easily made when dealing with 'virtual' objects like  $\lambda x. V_0^x(f)$ .

3.4.4. Advanced equivalences: Fourier series. We obtain an equivalence for  $NIN'_{alt}$  and properties of the Fourier series of BV-functions. Since the forward direction is rather involved, we have reserved a separate section for this result. Moreover, Theorem 3.22 is not at all straightforward: Jordan proves the convergence of Fourier series for BV-functions using the Jordan decomposition theorem, and the same for, e.g., [106, pp. 57–58]. However, the latter theorem seems much stronger<sup>15</sup> than  $NIN'_{alt}$ .

THEOREM 3.22 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN). The following are equivalent to NIN<sub>alt</sub><sup> $\prime$ </sup>.

- For  $f : [-\pi, \pi] \to \mathbb{R}$  in BV such that the Fourier coefficients exist, there is  $x_0 \in (-\pi, \pi)$  where the Fourier series  $S(f)(x_0)$  equals  $f(x_0)$ .
- For  $f : [-\pi, \pi] \to \mathbb{R}$  in BV such that the Fourier coefficients exist, the set of  $x \in (-\pi, \pi)$  where the Fourier series S(f)(x) equals f(x), is dense (or: not height countable, or: not countable, or: not strongly countable).

PROOF. Assume the first item and consider  $k : [0, 1] \to \mathbb{R}$  from (2.4). This function is Riemann integrable with  $\int_0^1 k(x) dx = 0$ , which one proves in the same way as for  $h : [0, 1] \to \mathbb{R}$  from (2.2) in the proof of Theorem 3.7. Similarly (and with a suitable rescaling), the Fourier coefficients of the Fourier series of k are zero. Hence, any  $x_0$  where the Fourier series of k converges to  $k(x_0)$  must be such that  $k(x_0) = 0$ , as required for NIN'<sub>alt</sub> since then  $x_0 \notin \bigcup_{n \in \mathbb{N}} X_n$ . Secondly, by items (iv)–(xix) in Theorem 3.20, NIN'<sub>alt</sub> implies that for a *BV*-

Secondly, by items (iv)–(xix) in Theorem 3.20, NIN'<sub>alt</sub> implies that for a *BV*-function, the set  $C_f$  is dense (or: not height countable, or: not countable, or: not strongly countable). Hence, the second item from Theorem 3.22 is immediate *if* we can show that S(f)(x) from Definition 3.19 equals  $\frac{f(x+)-f(x-)}{2}$  for *f* in *BV* and any *x* in the domain. Waterman provides an elementary and *almost* self-contained proof of this convergence fact in [101], avoiding the Jordan decomposition theorem and only citing [106, Vol. I, p. 55, (7.1)]. The proof of the latter is straightforward

<sup>&</sup>lt;sup>15</sup>The system  $\Pi_1^1$ -CA<sub>0</sub><sup> $\omega$ </sup> plus the Jordan decomposition theorem can prove  $\Pi_2^1$ -CA<sub>0</sub> [68]. By Theorem 3.15, WHBU implies NIN'<sub>alt</sub>, where the former seems weak in light of [66, Section 4].

trigonometry and Waterman's argument is readily formalised in ACA<sub>0</sub><sup> $\omega$ </sup>. Similarly, there are 'textbook' proofs that S(f)(x) equals  $\frac{f(x+)-f(x-)}{2}$  for f in BV and x in the domain that avoid the Jordan decomposition theorem. Such proofs generally seem to proceed as follows (see, e.g., [51, 102, 106]).

- By *Fejér's theorem* ([106, p. 89] or [102, p. 170]), Fourier series of *BV*-functions are convergent in the Césaro mean to <u>f(x+)-f(x-)</u>
   <u>2</u>.
- For *BV*-functions, Fourier coefficients are  $O(\frac{1}{n})$  ([102, p. 172], [106, p. 48]).
- By *Hardy's theorem* ([106, p. 78] or [102, p. 156]), if a series converges in the Césaro mean, it also converges in case the terms are  $O(\frac{1}{n})$ .
- By *Césaro's method of summation* (see [102, p. 155]), if a series converges in the Césaro mean to a limit *s* and the series also converges, then the series converges to the limit *s*.

Each of these results has an elementary (sometimes tedious and lengthy) proof that readily formalises in ACA<sub>0</sub><sup> $\omega$ </sup>. As an example, the proofs of the second item in [102, p. 172] and [1, p. 288] make use of the Jordan decomposition theorem, while the proofs in [92] and [106, p. 48] do not *and* are completely elementary. Finally, the perhaps 'most elementary' proof based on the above items is found in [51].  $\dashv$ 

Regarding the conditional nature of the items in Theorem 3.22, the Fourier coefficients of BV-functions of course exist by the Lebesgue criterion for the Riemann integral. However, that BV-functions have a point of continuity is already non-trivial by items (iv)–(xix) in Theorem 3.20. Moreover, the Darboux formulation of the Riemann integral involves suprema of BV-functions, which are hard to compute by the second cluster theorem in [71].

Finally, we could generalise the items from Theorems 3.20 and 3.22 to other notions of 'generalised' bounded variation. The latter notions yield (many) intermediate spaces between BV and regulated as follows.

**REMARK** 3.23 (Between bounded variation and regulated). The following spaces are intermediate between BV and regulated; all details may be found in [1].

Wiener spaces from mathematical physics [103] are based on *p*-variation, which amounts to replacing  $|f(x_i) - f(x_{i+1})|$  by  $|f(x_i) - f(x_{i+1})|^p$  in the definition of variation (1.2). Young [104] generalises this to  $\phi$ -variation which instead involves  $\phi(|f(x_i) - f(x_{i+1})|)$  for so-called Young functions  $\phi$ , yielding the Wiener-Young spaces. Perhaps a simpler construct is the Waterman variation [100], which involves  $\lambda_i |f(x_i) - f(x_{i+1})|$  and where  $(\lambda_n)_{n \in \mathbb{N}}$  is a sequence of reals with nice properties; in contrast to BV, any continuous function is included in the Waterman space [1, Proposition 2.23]. Combining ideas from the above, the Schramm variation involves  $\phi_i(|f(x_i) - f(x_{i+1})|)$  for a sequence  $(\phi_n)_{n \in \mathbb{N}}$  of well-behaved 'gauge' functions [85]. As to generality, the union (resp. intersection) of all Schramm spaces yields the space of regulated (resp. BV) functions, while all other aforementioned spaces are Schramm spaces [1, Propositions 2.43 and 2.46]. In contrast to BV and the Jordan decomposition theorem, these generalised notions of variation have no known 'nice' decomposition theorem. The notion of Korenblum variation [50] does have such a theorem (see [1, Proposition 2.68]) and involves a distortion function acting on the *partition*, not on the function values (see [1, Definition 2.60]).

It is no exaggeration to say that there are *many* natural spaces between the regulated and BV-functions, all of which yield equivalences in Theorem 3.20.

**3.5. When more is less in Reverse Mathematics.** An important-if not centralaspect of analysis is the relationship between its many function classes. It goes without saying that these relationships need not hold in weak logical systems. For instance, the well-known inclusion *regulated implies Baire 1* is not provable in  $Z_2^{\omega}$  + QF-AC<sup>0,1</sup> by [72, Theorem 2.34].

In this section, we establish a kind of dual to the previous negative result: we show that most of the above statements that are equivalent to NIN<sub>alt</sub> or NIN<sub>alt</sub>, become provable in the much weaker system  $ACA_0^{\omega} + QF-AC^{0,1} + FIN$  if we additionally require the functions to be Baire 1. To this end, we need the following theorem, where a *jump continuity* is any  $x \in (0, 1)$  such that  $f(x+) \neq f(x-)$ .

THEOREM 3.24 (ACA<sub>0</sub><sup> $\omega$ </sup>). If  $f : [0, 1] \to \mathbb{R}$  is regulated, there is a sequence of reals containing all jump discontinuities of f.

PROOF. This is immediate from [72, Theorem 2.16].

The following theorem should be contrasted with Theorems 3.8 and 3.20.

THEOREM 3.25 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN). For a Baire 1 function  $f : [0, 1] \rightarrow \mathbb{R}$  in BV, the points of continuity of f are dense.

**PROOF.** Let  $f : [0, 1] \to \mathbb{R}$  be Baire 1 and in *BV* with variation bounded by 1. This function is regulated by Theorem 3.3. Use Theorem 3.24 to enumerate the jump discontinuities of f as  $(y_n)_{n \in \mathbb{N}}$ . Let  $(f_n)_{n \in \mathbb{N}}$  be a sequence of continuous functions with pointwise limit f on [0, 1] and consider the following formula:

$$\varphi(n_0, k, x) \equiv (\forall n, m \ge n_0) (\forall q \in B(x, \frac{1}{2^m}) \cap \mathbb{Q}) (|f_n(x) - f(q)| \ge \frac{1}{2^k} + \frac{1}{2^{n_0}}).$$
(3.7)

For fixed  $k \in \mathbb{N}$ ,  $\varphi(n_0, k, x)$  holds for large enough  $n_0 \in \mathbb{N}$  in case f has a removable discontinuity at  $x \in (0, 1)$  such that  $|f(x) - f(x+)| > \frac{1}{2^k}$ . For fixed  $n_0, k \in \mathbb{N}$ , there can only be  $2^k$  many pairwise distinct  $x \in [0, 1]$  such that  $\varphi(n_0, k, x)$ , as each such real contributes at least  $\frac{1}{2^k}$  to the total variation of f.

Next, the formula  $\varphi(n_0, k, x)$  is equivalent to (second-order)  $\Pi_1^0$  as f only occurs with rational input and  $f_n$  can be replaced uniformly by a sequence of codes  $\Phi_n$ . Moreover, in case  $x =_{\mathbb{R}} y$ , then trivially  $\varphi(n_0, k, x) \leftrightarrow \varphi(n_0, k, y)$ , i.e., we have the extensionality property required for [89, II.5.7]. By the latter there is an RM-code of a closed set  $C_{n_0,k}$  such that  $x \in C_{n_0,k} \leftrightarrow \varphi(n_0, k, x)$  for all  $x \in \mathbb{R}$  and  $n_0, k \in \mathbb{N}$ . Since  $C_{n_0,k}$  is finite,  $O_{n_0,k} := [0, 1] \setminus (C_{n_0,k} \cup \{y_0, \dots, y_k\})$  is open and dense. By the Baire category theorem for RM-codes [89, II.5.8], the intersection  $\cap_{n_0,k \in \mathbb{N}} O_{n_0,k}$  is dense in [0, 1]. By definition, this intersection does not contain any points of discontinuity of f, and we are done.

The following corollary should be contrasted with Theorem 3.22.

1634

 $\dashv$ 

Corollary 3.26 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN).

- For a positive unordered sum  $\sum_{x \in [0,1]} a_x$  with upper bound and where  $(a_x)_{x \in [0,1]}$  is Baire 1, the set  $\{y \in [0,1] : a_y = 0\}$  is dense.
- For Baire 1 function  $f : [-\pi, \pi] \to \mathbb{R}$  in BV such that the Fourier coefficients exist, the set  $\{x \in (-\pi, \pi) : S(f)(x) = f(x)\}$ , is dense.

**PROOF.** Let  $(a_x)_{x \in [0,1]}$  be an unordered sum with upper bound 1. Now consider

$$X_n := \{ x \in [0, 1] : a_x > 1/2^n \},$$
(3.8)

which can have at most  $2^n$  elements. As in the proof of the theorem, the Baire 1 approximation of  $(a_x)_{x \in [0,1]}$  allows us to show that  $a_x > 1/2^n$  is (implied by) a suitable (second-order)  $\Sigma_1^0$ -formula. One then uses the (second-order) Baire category theorem to show that  $y \in [0, 1]$  such that  $a_y = 0$  are dense.

For the second item, following the proof of Theorem 3.22, f(x) = S(f)(x) holds in case f is continuous at  $x \in [0, 1]$ , where the latter is provided by the theorem.  $\dashv$ 

Next, the first item in Theorem 3.27 follows from [72, Theorem 2.26], but the latter is proved using  $ACA_0^{\omega} + ATR_0$ .

Theorem 3.27 (ACA<sub>0</sub><sup> $\omega$ </sup> + FIN + QF-AC<sup>0,1</sup>).

- For regulated Baire 1  $f : [0, 1] \to \mathbb{R}$ , the points of continuity of f are dense.
- *Volterra's corollary: there is no regulated and Baire* 1 *function that is continuous on*  $\mathbb{Q} \cap [0, 1]$  *and discontinuous on*  $[0, 1] \setminus \mathbb{Q}$ .
- For a Riemann integrable and regulated  $f : [0, 1] \rightarrow [0, 1]$  in Baire 1 with  $\int_0^1 f(x) dx = 0$ , the set  $\{x \in [0, 1] : f(x) = 0\}$  is dense.
- Blumberg's theorem [8] restricted to regulated Baire 1 functions on [0, 1].

**PROOF.** For the first item, the proof of Theorem 3.25 can be modified follows: for regulated f and fixed  $k \in \mathbb{N}$ , there can be at most *finitely many*  $x \in [0, 1]$  such that  $|f(x) - f(x+)| > \frac{1}{2^k}$ . One proves this fact by contradiction (as in the proof of Theorem 3.6), where QF-AC<sup>0,1</sup> provides a sequence  $(x_n)_{n \in \mathbb{N}}$  of reals in [0, 1] such that  $|f(x_n) - f(x_n+)| > \frac{1}{2^k}$  for all  $n \in \mathbb{N}$ . This sequence has a convergent sub-sequence, say with limit  $y \in [0, 1]$ ; one readily verifies that f(y+) or f(y-) does not exist. The rest of the proof is now identical to that of Theorem 3.25.

For the second item, the proof of Theorem 3.25 is readily adapted as follows: let  $(q_n)_{n \in \mathbb{N}}$  be an enumeration of the rationals in [0, 1] and define  $O'_{n_0,k}$  as  $O_{n_0,k} \setminus \{q_0, \ldots, q_k\}$ . Then Theorem 3.25 must yield an irrational point of continuity, a contradiction.

For the third item, note that f(x) = 0 must hold in case f is continuous at  $x \in [0, 1]$ , where the (dense set of the) latter is provided by the first item.

For the fourth item, this immediately follows from the first item.

 $\neg$ 

We could obtain similar results for most items of Theorem 3.7 and related theorems. We could also replace 'Baire 1' by 'effectively Baire 2' in Theorem 3.8 where the latter means that the function is given as the pointwise iterated limit of a double sequence of continuous functions; however, this would require us to go up to at least  $ATR_0$ , as suggested by the results in [72, Section 2.6].

Acknowledgements. We thank Anil Nerode and Chris Impens for their valuable advice. We also thank the anonymous referee for the helpful suggestions.

**Funding.** Our research was supported by the *Deutsche Forschungsgemeinschaft* via the DFG grant SA3418/1-1 and the *Klaus Tschira Boost Fund* via the grant Projekt KT43. Some results were obtained during the stimulating *Workshop on Reverse Mathematics and Philosophy* in June 2022 in Paris, sponsored by the NSF FRG grant *Collaborative Research: Computability Theoretic Aspects of Combinatorics.* We express our gratitude towards the aforementioned institutions.

## REFERENCES

[1] J. APPELL, J. BANA'S, and N. MERENTES, *Bounded Variation and Around*, De Gruyter Series in Nonlinear Analysis and Applications, vol. 17, De Gruyter, Berlin, 2014.

[2] J. AVIGAD and S. FEFERMAN, *Gödel's functional ("Dialectica") interpretation*, *Handbook of Proof Theory* (S. R. Buss, editor), Studies in Logic and the Foundations of Mathematics, vol. 137, Elsevier, Amsterdam, 1998, pp. 337–405.

[3] R. BAIRE, Sur les fonctions de variables réelles. Annali di Matematica, Serie III, vol. 3 (1899), no. 1, pp. 1–123.

[4] ——, *Leçons sur les Fonctions Discontinues*, Les Grands Classiques Gauthier-Villars, Éditions Jacques Gabay, Sceaux, 1995 (in French), reprint of the 1905 original.

[5] J. M. BARRETT, *The reverse mathematics of Cousin's lemma*, Honour's thesis, Victoria University of Wellington, 2020, supervised by Rod Downey and Noam Greenberg, xi + 51 pp.

[6] J. M. BARRETT, R. G. DOWNEY, and N. GREENBERG, *Cousin's lemma in second-order arithmetic*, Preprint, 2021, arXiv: https://arxiv.org/abs/2105.02975.

[7] E. BISHOP, Foundations of Constructive Analysis, McGraw-Hill, New York, 1967.

[8] H. BLUMBERG, New properties of all real functions. Transactions of the American Mathematical Society, vol. 24 (1922), no. 2, pp. 113–128.

[9] N. BOURBAKI, Élements de mathématique, livre IV: Fonctions d'une variable rélle (Théorie élémentaire), Scientifiques et Industrielles, vol. 1132, Hermann et Cie, Paris, 1951 (in French).

[10] ——, *Functions of a Real Variable*, Elements of Mathematics, Springer, Berlin-Heidelberg, 2004.

[11] D. BRIDGES, A constructive look at functions of bounded variation. The Bulletin of the London Mathematical Society, vol. 32 (2000), no. 3, pp. 316–324.

[12] D. BRIDGES and A. MAHALANOBIS, Bounded variation implies regulated: A constructive proof, this JOURNAL, vol. 66 (2001), no. 4, pp. 1695–1700.

[13] D. S. BRIDGES, *Constructive Functional Analysis*, Research Notes in Mathematics, vol. 28, Pitman, London–San Francisco–Melbourne, 1979.

[14] ——, Reflections on function spaces. Annals of Pure and Applied Logic, vol. 163 (2012), no. 2, pp. 101–110.

[15] D. K. BROWN and S. G. SIMPSON, Which set existence axioms are needed to prove the separable Hahn–Banach theorem? Annals of Pure and Applied Logic, vol. 31 (1986), nos. 2–3, pp. 123–144.

[16] J. B. BROWN, A measure theoretic variant of Blumberg's theorem. Proceedings of the American Mathematical Society, vol. 66 (1977), no. 2, pp. 266–268.

[17] A. M. BRUCKNER, *Differentiation of Real Functions*, Lecture Notes in Mathematics, vol. 659, Springer, Berlin–Heidelberg, 1978.

[18] W. BUCHHOLZ, S. FEFERMAN, W. POHLERS, and W. SIEG, *Iterated Inductive Definitions and Subsystems of Analysis*, Lecture Notes in Mathematics, vol. 897, Springer, Berlin–Heidelberg, 1981.

[19] G. CANTOR, Ueber eine Eigenschaft des Inbegriffs Aller reellen algebraischen Zahlen. Journal für die Reine und Angewandte Mathematik, vol. 77 (1874), pp. 258–262.

[20] J. L. COOLIDGE, The lengths of curves. American Mathematical Monthly, vol. 60 (1953), pp. 89–93.

[21] P. COUSIN, Sur les fonctions de n variables complexes. Acta Mathematica, vol. 19 (1895), pp. 1–61.

[22] G. DARBOUX, Mémoire Sur les fonctions discontinues. Annales Scientifiques de l'École Normale Supérieure 2e série, vol. 4 (1875), pp. 57–112.

[23] L. P. G. DIRICHLET, Über die Darstellung ganz willkürlicher Funktionen durch Sinusund Cosinusreihen, Repertorium der physik, von H.W. Dove und L. Moser, bd. 1, 1837.

[24] E. DIRKSEN, Ueber die Anwendung der Analysis auf die Rectification der Curven, Akademie der Wissenschaften zu Berlin (1833), pp. 123–168.

[25] J. M. C. DUNHAM, Des méthodes dans les sciences de raisonnement. Application des méthodes générales à la science des nombres et à la science de l'étendue, vol. II, Gauthier-Villars, 1866.

[26] D. D. DZHAFAROV and C. MUMMERT, *Reverse Mathematics: Problems, Reductions, and Proofs*, Springer, Cham, 2022.

[27] H. W. ELLIS, Darboux properties and applications to non-absolutely convergent integrals. Canadian Journal of Mathematics, vol. 3 (1951), pp. 471–485.

[28] H. FRIEDMAN, Some systems of second order arithmetic and their use, Proceedings of the International Congress of Mathematicians, vol. 1 (R. D. James, editor), Canadian Mathematical Congress, Vancouver, 1975, pp. 235–242.

[29] — , Systems of second order arithmetic with restricted induction, I & II (abstracts), this Journal, vol. 41 (1976), pp. 557–559.

[30] D. GAULD, Did the Young Volterra know about Cantor? Mathematics Magazine, vol. 66 (1993), no. 4, pp. 246–247.

[31] N. GREENBERG, J. S. MILLER, and A. NIES, *Highness properties close to PA completeness*. Israel Journal of Mathematics, vol. 244 (2019), pp. 419–465.

[32] F. HAUSDORFF, Problème 62. Fundamenta Mathematicae, vol. 25 (1935), p. 478.

[33] A. HEYTING, *Recent progress in intuitionistic analysis*, *Intuitionism and Proof Theory* (R. E. Vesley, A. Kino, and J. Myhill, editors), North-Holland, Amsterdam, 1970, pp. 95–100.

[34] D. HILBERT, Über das Unendliche. Mathematische Annalen, vol. 95 (1926), no. 1, pp. 161–190 (in German).

[35] D. HILBERT and P. BERNAYS, *Grundlagen der Mathematik. II*, Zweite Auflage. Die Grundlehren der Mathematischen Wissenschaften, Band 50, Springer, 1970.

[36] K. HRBACEK and T. JECH, *Introduction to Set Theory*, third ed., Monographs and Textbooks in Pure and Applied Mathematics, vol. 220, Marcel Dekker, New York, 1999.

[37] F. N. HUGGINS, Some interesting properties of the variation function. American Mathematical Monthly, vol. 83 (1976), no. 7, pp. 538–546.

[38] J. HUNTER, *Higher-order reverse topology*, Ph.D. thesis, The University of Wisconsin–Madison, 2008.

[39] J. K. HUNTER and B. NACHTERGAELE, Applied Analysis, World Scientific, River Edge, 2001.

[40] M. N. HUXLEY, *Area, Lattice Points, and Exponential Sums*, London Mathematical Society Monographs. New Series, vol. 13, The Clarendon Press, Oxford University Press, Oxford Science Publications, New York, 1996.

[41] J. E. JAYNE and C. A. ROGERS, First level Borel functions and isomorphisms, Journal de Mathématiques Pures et Appliquées (9), vol. 61 (1982), no. 2, pp. 177–205.

[42] C. JORDAN, Sur la série de Fourier. Comptes Rendus de l'Académie des Sciences, vol. 92 (1881), pp. 228–230.

[43] ——, *Cours d'analyse de L' École Polytechnique*, *Tome I*, les Grands Classiques Gauthier-Villars, Éditions Jacques Gabay, Sceaux, 1991. Reprint of the third (1909) edition; first edition: 1883.

[44] J. L. KELLEY, *General Topology*, Graduate Texts in Mathematics, vol. 27, Springer, New York, 1975, reprint of the 1955 edition.

[45] B. KIRCHHEIM, Baire one star functions. Real Analysis Exchange, vol. 18 (1992/93), no. 2, pp. 385–399.

[46] S. C. KLEENE, Recursive functionals and quantifiers of finite types. I. Transactions of the American Mathematical Society, vol. 91 (1959), pp. 1–52.

[47] U. KOHLENBACH, Foundational and mathematical uses of higher types, Reflections on the Foundations of Mathematics (W. Sieg, R. Sommer, and C. Talcott, editors), Lecture Notes in Logic, vol. 15, ASL, Urbana, 2002, pp. 92–116.

[48] — *Higher order reverse mathematics, Reverse Mathematics 2001* (S. G. Stephen, editor), Lecture Notes in Logic, vol. 21, ASL, Urbana, 2005, pp. 281–295.

[49] A. N. KOLMOGOROV and S. V. FOMIN, *Elements of the Theory of Functions and Functional Analysis. Vol. 1. Metric and Normed Spaces*, Graylock Press, Rochester, 1957, translated from the first Russian edition by Leo F. Boron.

[50] B. KORENBLUM, An extension of the Nevanlinna theory. Acta Mathematica, vol. 135 (1975), nos. 3–4, pp. 187–219.

[51] T. W. KÖRNER, Fourier Analysis, Cambridge University Press, Cambridge, 1988.

[52] G. KOUMOULLIS, A generalization of functions of the first class. Topology and Its Applications, vol. 50 (1993), no. 3, pp. 217–239.

[53] A. P. KREUZER, Bounded variation and the strength of Helly's selection theorem. Logical Methods in Computer Science, vol. 10 (2014), no. 4, p. 4:16, 15.

[54] K. KUNEN, Set Theory, Studies in Logic, vol. 34, College Publications, London, 2011.

[55] K. KURATOWSKI, *Topology*, vol. I, Academic Press, New York–London, 1966.

[56] I. LAKATOS, *Proofs and Refutations*, Cambridge Philosophy Classics, Cambridge University Press, Cambridge, 2015. The logic of mathematical discovery; Edited by John Worrall and Elie Zahar; with a new preface by Paolo Mancosu; originally published in 1976.

[57] P.-Y. LEE, W.-K. TANG, and D. ZHAO, An equivalent definition of functions of the first Baire class. *Proceedings of the American Mathematical Society*, vol. 129 (2001), no. 8, pp. 2273–2275.

[58] J. LONGLEY and D. NORMANN, *Higher-Order Computability*, Theory and Applications of Computability, Springer, Berlin–Heidelberg, 2015.

[59] V. H. MOLL, *Numbers and Functions*, Student Mathematical Library, vol. 65, American Mathematical Society, Providence, 2012.

[60] A. MONTALBÁN, *Open questions in reverse mathematics*. *Bulletin of Symbolic Logic*, vol. 17 (2011), no. 3, pp. 431–454.

[61] E. H. MOORE, Definition of limit in general integral analysis. Proceedings of the National Academy of Sciences of the United States of America, vol. 1 (1915), no. 12, pp. 628–632.

[62] E. H. MOORE and H. SMITH, A general theory of limits. American Journal of Mathematics, vol. 44 (1922), pp. 102–121.

[63] G. I. MYERSON, *First-class functions*. *American Mathematical Monthly*, vol. 98 (1991), no. 3, pp. 237–240.

[64] A. NIES, M. A. TRIPLETT, and K. YOKOYAMA, *The reverse mathematics of theorems of Jordan and Lebesgue*, this JOURNAL, vol. 86 (2021), pp. 1657–1675.

[65] D. NORMANN and S. SANDERS, On the mathematical and foundational significance of the uncountable. Journal of Mathematical Logic, vol. 19 (2019), no. 1, p. 1950001.

[66] ——, *The Vitali covering theorem in computability theory and reverse mathematics*, submitted, 2019, arXiv: https://arxiv.org/abs/1902.02756.

[67] ——, Pincherle's theorem in reverse mathematics and computability theory. Annals of Pure and Applied Logic, vol. 171 (2020), no. 5, p. 102788, 41 pp.

[68] ———, On robust theorems due to Bolzano, Jordan, Weierstrass, and Cantor in reverse mathematics, this JOURNAL (2022), pp. 1– 51. https://doi.org/10.1017/jsl.2022.71

[69] —, On the uncountability of  $\mathbb{R}$ , this JOURNAL (2022), pp. 1–43. https://doi.org/10.1017/ jsl.2022.27

[70] ——, *Betwixt Turing and Kleene*, Lecture Notes in Computer Science, vol. 13137, Proceedings of LFCS22, 2022, p. 18.

[71] ——, On the computational properties of basic mathematical notions. Journal of Logic and Computation (2022), pp. 1–44. https://doi.org/10.1093/logcom/exac075

[72] ——, The biggest five of reverse mathematics, submitted, 2022, pp. 1–38. arXiv: https://arxiv.org/abs/2212.00489

[73] F. RICHMAN, Omniscience principles and functions of bounded variation. Mathematical Logic Quarterly, vol. 48 (2002), pp. 111–116.

[74] B. RIEMANN, Ueber die Darstellbarkeit einer Function durch eine trigonometrische Reihe, Abhandlungen der Königlichen Gesellschaft der Wissenschaften zu Göttingen, vol. 13. Habilitation thesis defended in 1854, published in 1867, p. 47.

[75] H. L. ROYDEN, Real Analysis, Lecture Notes in Mathematics, Pearson Education, London, 1989.

[76] N. SAKAMOTO and T. YAMAZAKI, Uniform versions of some axioms of second order arithmetic, Mathematical Logic Quarterly, vol. 50 (2004), no. 6, pp. 587–593.

[77] S. SANDERS, Reverse mathematics of the uncountability of  $\mathbb{R}$ : Baire classes, metric spaces, and unordered sums, submitted, 2020, p. 15. arXiv: https://arxiv.org/abs/2011.02915

[78] ——, *Countable sets versus sets that are countable in reverse mathematics. Computability*, vol. 11 (2022), no. 1, pp. 9–39.

[79] —, The Uncountability of  $\mathbb{R}$  in Reverse Mathematics, Lecture Notes in Computer Science, vol. 13359, Proceedings of CiE22, Springer, 2022, pp. 272-286.

[80] ——, On the computational properties of the uncountability of the real numbers, Logic, Language, Information, and Computation (A. Ciabattoni, E. Pimentel, and R. J. G. B. de Queiroz, editors), Lecture Notes in Computer Science, vol. 13468, Springer, Cham, 2022, pp. 362-377. https://doi.org/10.1007/978-3-031-15298-6\_23

[81] ——, Finding points of continuity in real analysis, submitted, 2022.
[82] ——, On the computational properties of the Baire category theorem, submitted, 2022, arXiv: https://arxiv.org/abs/2210.05251.

[83] ——, Big in reverse mathematics: Measure and category, submitted, 2023, arXiv: https://arxiv.org/abs/2303.00493.

[84] L. SCHEEFFER, Allgemeine Untersuchungen über rectification der Curven. Acta Mathematica, vol. 5 (1884), no. 1, pp. 49–82 (in German).

[85] M. SCHRAMM, Functions of  $\Phi$ -bounded variation and Riemann–Stieltjes integration. Transactions of the American Mathematical Society, vol. 287 (1985), no. 1, pp. 49-63.

[86] C. E. SILVA and Y. WU. No functions continuous only at points in a countable dense set, preprint, 2018, arXiv: https://arxiv.org/abs/1809.06453v3.

[87] S. G. SIMPSON, Which set existence axioms are needed to prove the Cauchy/Peano theorem for ordinary differential equations? this JOURNAL, vol. 49 (1984), no. 3, pp. 783–802.

[88] ——, editor, *Reverse Mathematics 2001*, Lecture Notes in Logic, vol. 21, ASL, La Jolla, 2005.

[89] ——, Subsystems of Second Order Arithmetic, second ed., Perspectives in Logic, Cambridge University Press, Cambridge, 2009.

[90] H. H. SOHRAB, Basic Real Analysis, second ed., Birkhäuser/Springer, New York, 2014.

[91] J. STILLWELL, Reverse Mathematics, Proofs from the Inside Out, Princeton University Press, Princeton, 2018.

[92] M. TAIBLESON, Fourier coefficients of functions of bounded variation. Proceedings of the American Mathematical Society, vol. 18 (1967), p. 766.

[93] T. TAO, An Epsilon of Room, I: Real Analysis, Graduate Studies in Mathematics, vol. 117, American Mathematical Society, Providence, 2010.

[94] — ----, An Introduction to Measure Theory, Graduate Studies in Mathematics, vol. 126, American Mathematical Society, Providence, 2011.

[95] C. J. T. THOMAE, *Einleitung in Die Theorie der Bestimmten Integrale*, Louis Nebert, Halle, 1875.

[96] J. W. TUKEY, Convergence and Uniformity in Topology, Annals of Mathematics Studies, vol. 2, Princeton University Press, Princeton, 1940.

[97] B. S. VATSSA, Discrete Mathematics, fourth ed., New Age International, New Delhi, 1993.

[98] V. VOLTERRA, Alcune osservasioni sulle funzioni punteggiate discontinue. Giornale di Matematiche, vol. XIX (1881), pp. 76-86.

[99] F. WAALDIJK, On the foundations of constructive mathematics—Especially in relation to the theory of continuous functions. Foundations of Science, vol. 10 (2005), no. 3, pp. 249–324.

[100] D. WATERMAN, On convergence of Fourier series of functions of generalized bounded variation. Studia Mathematica, vol. 44 (1972), pp. 107-117.

-, Bounded variation and Fourier series. Real Analysis Exchange, vol. 3 (1977/78), no. 2, [101] pp. 61–85.

[102] E. T. WHITTAKER and G. N. WATSON, A Course of Modern Analysis, Cambridge Mathematical Library, Cambridge University Press, Cambridge, 1996. An introduction to the general theory of infinite processes and of analytic functions; with an account of the principal transcendental functions; reprint of the fourth (1927) edition.

[103] N. WIENER. The quadratic variation of a function and its Fourier coefficients. Journal of Mathematics and Physics, vol. 3 (1924), pp. 73-94.

[104] L. C. YOUNG, Sur Une généralisation de la notion de variation de puissance pième bornée au Sens de Wiener, et Sur la convergence des séries de Fourier. Comptes Rendus de l'Académie des Sciences, vol. 204 (1937), 470–472.

[105] X. ZHENG and R. RETTINGER, Effective Jordan decomposition. Theory of Computing Systems, vol. 38 (2005), no. 2, pp. 189–209.

[106] A. ZYGMUND, Trigonometric Series, vols. I and II, second ed., Cambridge University Press, New York, 1959.

1640

DEPARTMENT OF PHILOSOPHY II RUB BOCHUM BOCHUM, GERMANY *E-mail*: sasander@me.com